

MIDI p.l.c.

a public limited company incorporated under the laws of Malta, company registration number C 15836

Issue of:

€50,000,000 4% Secured Bonds 2026 ISIN: MT0000421223

SECURITIES NOTE

Dated 28 June 2016

This document is a Securities Note issued pursuant to the requirements of Rule 4.14 of the Listing Rules of the Listing Authority and Commission Regulation (EC) No. 809/2004 of 29 April 2004 implementing Directive 2003/71/EC of the European Parliament and of the Council as regards information contained in prospectuses as well as the format, incorporation by reference and publication of such prospectuses and dissemination of advertisements, as subsequently amended. This Securities Note should be read in conjunction with the Registration Document containing information about the Issuer dated 28 June 2016.

THE AUTHORISATION BY THE LISTING AUTHORITY FOR THE ADMISSIBILITY OF THESE SECURITIES AS A LISTED FINANCIAL INSTRUMENT MEANS THAT THE SAID INSTRUMENTS ARE IN COMPLIANCE WITH THE REQUIREMENTS AND CONDITIONS SET OUT IN THE LISTING RULES. IN PROVIDING THIS AUTHORISATION, THE LISTING AUTHORITY DOES NOT GIVE ANY CERTIFICATION REGARDING THE POTENTIAL RISKS IN INVESTING IN THE SAID INSTRUMENT AND SUCH AUTHORISATION SHOULD NOT BE DEEMED OR BE CONSTRUED AS A REPRESENTATION OR WARRANTY AS TO THE SAFETY OF INVESTING IN SUCH INSTRUMENT.

THE LISTING AUTHORITY ACCEPTS NO RESPONSIBILITY FOR THE CONTENTS OF THE PROSPECTUS, MAKES NO REPRESENTATIONS AS TO ITS ACCURACY OR COMPLETENESS AND EXPRESSLY DISCLAIMS ANY LIABILITY WHATSOEVER FOR ANY LOSS HOWEVER ARISING FROM OR IN RELIANCE UPON THE WHOLE OR ANY PART OF THE CONTENTS OF THE PROSPECTUS INCLUDING ANY LOSSES INCURRED BY INVESTING IN THESE SECURITIES.

A PROSPECTIVE INVESTOR SHOULD ALWAYS SEEK INDEPENDENT FINANCIAL ADVICE BEFORE DECIDING TO INVEST IN ANY LISTED FINANCIAL INSTRUMENTS. A PROSPECTIVE INVESTOR SHOULD BE AWARE OF THE POTENTIAL RISKS IN INVESTING IN THE SECURITIES OF AN ISSUER AND SHOULD MAKE THE DECISION TO INVEST ONLY AFTER CAREFUL CONSIDERATION AND CONSULTATION WITH HIS OR HER OWN INDEPENDENT FINANCIAL ADVISER.

David Demarco

Mizzi

Legal Advisers

David G. Curmi

Security Trustee

Mark Portelli Joseph Said

Manager & Registrai

MAMO TCV

,

Sponsor

BOV Bank of Valletta

eph Bonello

CSB GROU

ECHARTS
WEALTH MANAGEMENT - CORPORATE BROKING

COSCUMITION NO

IMPORTANT INFORMATION

This Securities Note, forming part of the Prospectus, contains information on an issue by the Issuer, MIDI p.l.c., of €50,000,000 Secured Bonds due 2026 having a nominal value of €100. The Bonds will be issued at par and bear interest at the rate of 4% per annum payable annually on 27 July of each year, until the Redemption Date. The nominal value of the Bonds will be repayable in full at maturity on the Redemption Date unless otherwise previously repurchased for cancellation.

This Securities Note contains information about the Issuer and the Bonds in accordance with the requirements of the Listing Rules, the Companies Act, and the Commission Regulation.

No broker, dealer, salesman or other person has been authorised by the Issuer or its directors, to issue any advertisement or to give any information or to make any representations in connection with the sale of Bonds of the Issuer other than those contained in the Prospectus and in the documents referred to herein, and if given or made, such information or representations must not be relied upon as having been authorised by the Issuer or its Directors or advisers.

The Prospectus does not constitute, and may not be used for purposes of, an offer or invitation to subscribe for Bonds by any person in any jurisdiction (i) in which such offer or invitation is not authorised or (ii) in which the person making such offer or invitation is not qualified to do so or (iii) to any person to whom it is unlawful to make such offer or invitation.

The Prospectus and the offering, sale or delivery of any securities may not be taken as an implication: (i) that the information contained in the Prospectus is accurate and complete subsequent to its date of issue; or (ii) that there has been no material adverse change in the financial position of the Issuer since such date; or (iii) that any other information supplied in connection with the Prospectus is accurate at any time subsequent to the date on which it is supplied or, if different, the date indicated in the document containing the same.

It is the responsibility of any persons in possession of this document and any persons wishing to apply for any securities issued by the Issuer to inform themselves of, and to observe and comply with, all applicable laws and regulations of any relevant jurisdiction. Prospective applicants for any securities that may be issued by the Issuer should inform themselves as to the legal requirements of applying for any such securities and any applicable exchange control requirements and taxes in the countries of their nationality, residence or domicile.

Save for the issue in the Republic of Malta, no action has been or will be taken by the Issuer that would permit a public offering of the Bonds or the distribution of the Prospectus (or any part thereof) or any offering material in any country or jurisdiction where action for that purpose is required. In relation to each member state of the European Economic Area (other than Malta) which has implemented Directive 2003/71/EC of the European Parliament and of the Council of 4 November 2003 on the prospectus to be published when securities are offered to the public or admitted to trading, as subsequently amended, or which, pending such implementation, applies article 3.2 of said Directive, the Bonds can only be offered to "Qualified Investors" (as defined in said Directive) as well as in any other circumstances which do not require the publication by the Issuer of a prospectus pursuant to Article 3 of said Directive.

The Bonds have not been nor will they be registered under the United States Securities Act, 1933 as amended, or under any federal or state securities law and may not be offered, sold or otherwise transferred, directly or indirectly, in the United States of America, its territories or possessions, or any area subject to its jurisdiction (the "U.S.") or to or for the benefit of, directly or indirectly, any U.S. person (as defined in Regulation "S" of the said Act). Furthermore the Issuer will not be registered under the United States Investment Company Act, 1940 as amended and investors will not be entitled to the benefits set out therein.

All the advisers to the Issuer named under the heading "Advisers to the Issuer" and "Statutory Auditors and Financial Advisers" in sections 4.1 and 4.2 respectively of this Securities Note have acted and are acting exclusively for the Issuer in relation to this issue and have no contractual, fiduciary or other obligation towards any other person and will accordingly not be responsible to any investor or any other person whomsoever in relation to the transactions proposed in the Prospectus.

The contents of the Issuer's website or any website directly or indirectly linked to the Issuer's website do not form part of the Prospectus. Accordingly no reliance ought to be made by any investor on any information or other data contained in such websites as the basis for a decision to invest in the Bonds.

Prospective investors should carefully consider all the information contained in the prospectus as a whole and should consult their own independent financial and other professional advisers before deciding to make an investment in the Bonds.

The value of investments can go up or down and past performance is not necessarily indicative of future performance. The nominal value of the Bonds will be repayable in full upon maturity.

Statements made in the Prospectus are, except where otherwise stated, based on the law and practice currently in force in Malta and are subject to changes therein.

A copy of this document has been submitted to the Listing Authority in satisfaction of the Listing Rules, the Malta Stock Exchange in satisfaction of the Malta Stock Exchange Bye-Laws and has been duly filed with the Registrar of Companies, in accordance with the Companies Act.

INFORMATION REGARDING CONSENT BY THE ISSUER AND THE DIRECTORS TO THE USE OF THE PROSPECTUS IN THE EVENT OF ANY RESALE OR FINAL PLACEMENT OF BONDS BY AUTHORISED FINANCIAL INTERMEDIARIES

The Issuer and the Directors consent to the use of this Prospectus (and accept responsibility for the information contained therein) with respect to any subsequent resale or final placement of Bonds by any Authorised Financial Intermediaries, in circumstances where there is an offer of securities to the public which requires the publication of a prospectus in terms of the Prospectus Directive, provided this is limited only:

- i. to any resale or final placement of Bonds taking place in Malta; and
- ii. to any resale or final placement of Bonds taking place within the period of 60 days from the date of the Prospectus.

Neither the Issuer nor the Directors nor the Sponsor have any responsibility for any of the actions of any Authorised Financial Intermediary, including their compliance with applicable conduct of business rules or other local regulatory requirements or other securities law requirements in relation to a resale or final placement of Bonds.

Other than as set out above, neither the Issuer nor the Directors nor the Sponsor have authorised (nor do they authorise or consent to the use of this Prospectus in connection with) the making of any public offer of the Bonds by any person in any circumstances. Any such unauthorised offers are not made on behalf of the Issuer or the Directors or the Sponsor and neither the Issuer nor the Directors nor the Sponsor has any responsibility or liability for the actions of any person making such offers.

Investors should enquire whether an intermediary is considered to be an Authorised Financial Intermediary in terms of the Prospectus. If the investor is in doubt as to whether it can rely on the Prospectus and/or who is responsible for its contents, it should seek legal advice.

No person has been authorised to give any information or to make any representation not contained in or inconsistent with this Prospectus. If given or made, it must not be relied upon as having been authorised by the Issuer or the Directors or Sponsor. The Issuer and the Directors do not accept responsibility for any information not contained in this Prospectus.

In the event of a resale or final placement of Bonds by an Authorised Financial Intermediary, the Authorised Financial Intermediary shall be responsible to provide information to investors on the terms and conditions of the resale or final placement at the time such is made.

Any resale or final placement of Bonds to an investor by an Authorised Financial Intermediary, will be made in accordance with any terms and other arrangements in place between such Authorised Financial Intermediary and such investor including as to price, allocations and settlement arrangements. Where such information is not contained in the Prospectus, it will be the responsibility of the applicable Authorised Financial Intermediary at the time of such resale or final placement to provide the investor with that information and neither the Issuer nor the Sponsor has any responsibility or liability for such information.

Any Authorised Financial Intermediary using this Prospectus in connection with a resale or final placement of Bonds subsequent to the Bond Issue shall from the date of the Prospectus, publish on its website a notice to the effect that it is using this Prospectus for such resale or final placement in accordance with the consent of the Issuer and the Directors and the conditions attached thereto.

Any new information with respect to Authorised Financial Intermediaries unknown at the time of the approval of this Securities Note will be made available through a company announcement which will also be made available on the Issuer's website: www.midimalta.com

The consent provided herein shall no longer apply following the lapse of a period of 60 days from the date of the Prospectus.

FORWARD LOOKING STATEMENTS

This Securities Note may contain "forward looking statements" which include, among others, statements concerning matters that are not historical facts and which may involve projections of future circumstances. These forward looking statements are subject to a number of risks, uncertainties and assumptions and important factors that could cause actual risks to differ materially from the expectations of the Issuer's Directors. No assurance is given that future results or expectations will be achieved.

Table of Contents

Imp	ortant Information	2
	ormation regarding consent by the Issuer and the Directors to the use of the Prospectus he event of any Resale or Final Placement of Bonds by Authorised Financial Intermediaries	3
For	ward Looking Statements	4
1.	Definitions	7
2.	Risk Factors 2.1 Risks relating to the Bonds 2.2 Risks relating to the Security Interest	9 10 10
3.	Persons Responsible	11
4.	Advisers to the Issuer, Auditors and Security Trustee 4.1 Advisers to the Issuer 4.2 Statutory Auditors and Financial Advisers 4.3 Security Trustee	11 11 12 12
5.	Essential Information 5.1 Interest of Natural and Legal Persons involved in the Offer 5.2 Reasons for the Offer and Use of Proceeds 5.3 Expenses	12 12 12 13
6.	Offer Statistics	14
7.	Expected Timetable	15
8.	Details of the Offer and Admission to Trading 8.1 The Bond Issue 8.2 Distribution and Offer Period 8.3 Allocation Policy 8.4 Allotment Results 8.5 Bond Exchange Programme 8.6 Refunds 8.7 Admission to Trading	16 16 16 16 17 17 18
9. Ir	9.1 Description of the Bonds 9.2 Currency of the Bond Issue 9.3 Rights attached to the Bonds 9.4 Summary of Existing Registered Privileges and Hypothecs 9.5 The Security Interest 9.6 The Trust Instrument 9.7 Registration, Form, Denomination and Title 9.8 Interest 9.9 Transferability of the Bonds 9.10 Maturity and Redemption 9.11 Yield 9.12 Authorisations and Approval 9.13 Expected Issue Date of the Bonds	18 18 19 19 19 22 24 25 25 25 25
10.	Taxation 10.1 General 10.2 Malta Tax on Interest	25 25 26

10.3 Exchange of Information 10.4 Maltese Taxation on Capital Gains on Transfer of the Bonds 10.5 Duty on Documents and Transfers	26 26 26
11. Terms and Conditions of the Bonds 11.1 General 11.2 Form, Denomination, Title and Pricing 11.3 Status and Security Interest 11.4 Interest 11.5 Payments 11.6 Redemption and Purchase 11.7 Events of Default 11.8 Registration, Replacement, Transfer and Exchange 11.9 Resolutions and Meetings of Bondholders 11.10 Further Issues 11.11 Bonds held Jointly 11.12 Bonds held Subject to Usufruct 11.13 Governing Law and Jurisdiction 11.14 Notices 11.15 Listing	27 27 27 28 28 28 29 30 31 32 32 32 32 33
12. Terms and Conditions of the Application	33
Annex I – Specimen Application Forms	37
Annex II – Authorised Financial Intermediaries	46
Annex III – Summary of Privileges and Hypothecs	48
Annex IV - Financial Analysis Summary	FAS 1

1. DEFINITIONS

Words and expressions and capitalised terms used in this Securities Note shall, except where the context otherwise requires and except where otherwise defined herein, bear the same meaning as the meaning given to such words, expressed and capitalised terms as indicated in the Registration Document issued by the Issuer on 28 June 2016.

Applicant	a person or persons whose name or names (in the case of joint applicants) appear in the registration details of an Application Form;
Application/s	the application to subscribe for Bonds made by an Applicant/s by completing an Application Form/s and delivering same to the Issuer or to any of the other Authorised Financial Intermediaries;
Application Form	the forms of application for the subscription for the Bonds, specimens of which are contained in Annex I of this Securities Note;
Authorised Financial Intermediaries	the licensed stockbrokers and financial intermediaries listed in Annex II of this Securities Note;
Bond(s) or Secured Bond(s)	the four per cent (4%) secured bonds due 27 July 2026 being issued pursuant to the Prospectus having a nominal value of €100 each for an aggregate principal amount of fifty million euro (€50,000,000);
Bonds 2016/18	the EUR Bonds 2016/18 and GBP Bonds 2016/18;
Bond Conditions	the terms and conditions applicable to the Bonds set out in section 11 under the heading "Terms and Conditions of the Bonds" of this Securities Note;
Bond Exchange Programme	the bond exchange programme set out in section 8.5 under the heading "Bond Exchange Programme" of this Securities Note;
Bondholder	a holder of the Bonds;
Bond Issue	the issue of the Bonds;
Bond Issue Price or Issue Price	the price of €100 per Bond;
Business Day	any day between Monday and Friday (both days included) on which commercial banks in Malta settle payments and are open for normal banking business;
Commission Regulation	Commission Regulation (EC) No. 809/2004 of 29 April 2004 implementing Directive 2003/71/EC of the European Parliament and of the Council as regards information contained in prospectuses as well as the format incorporation by reference and publication of such prospectuses and dissemination of advertisements, as subsequently amended;
Companies Act	the Companies Act (Cap. 386 of the laws of Malta);
Company; Issuer or MIDI	MIDI p.l.c., a company registered in Malta with registration number C 15836;
CSD	the Central Securities Depository of the Malta Stock Exchange, having its address at Malta Stock Exchange, Garrison Chapel, Castille Place, Valletta, VLT 1063, Malta;
Current Shareholders	shareholders of the Issuer as at the Cut-Off Date;
Cut-Off Date	close of business on 24 June 2016 (trading session of 22 June 2016);
Emphyteutical Deed	the public deed in the records of Notary Vincent Miceli of 15 June 2000 whereby the GOM, acting through the Land Department granted the Company the Emphyteutical Grant;

Emphyteutical Grant	the temporary emphyteutical concession of the Emphyteutical Land for a period of ninety nine years commencing from 15 June 2000 made by GOM to the Company by virtue of the Emphyteutical Deed;
Emphyteutical Land	the immovable property comprising Tigné Point and Manoel Island forming the subject-matter of the Emphyteutical Grant;
EUR Bonds 2016/18	the \le 31,702,900 7% bonds 2016-2018 (ISIN: MT0000421207) issued by the Issuer pursuant to a prospectus dated 5 December 2008;
Euro or €	the lawful currency of the Republic of Malta;
Existing Holders	existing holders of the Bonds 2016/18 as held on the Cut-Off Date;
GBP Bonds	the £7,214,300 7% bonds 2016-2018 (ISIN: MT0000421215) issued by the Issuer pursuant
2016/18	to a prospectus dated 5 December 2008;
GOM	the Government of Malta;
Group or MIDI Group	the Issuer and the subsidiary companies of the Issuer and the term "Group Company" shall be construed accordingly;
Initial Security Interest	has the meaning set out in section 9.5 of this Securities Note;
Interest Payment Date	27 July of each year, between 2017 and the year in which the Bonds are redeemed (both years included), provided that any Interest Payment Date which falls on a day other than a Business Day will be carried over to the next following day that is a Business Day;
Issue Date	expected on 03 August 2016;
Listing Authority	the Malta Financial Services Authority, as appointed in terms of the Financial Markets Act (Cap. 345 of the laws of Malta);
Listing Rules	the listing rules issued by the Listing Authority;
Malta Stock Exchange or MSE	the Malta Stock Exchange p.l.c., as originally constituted in terms of the Financial Markets Act (Cap. 345 of the laws of Malta) with company registration number C 42525 and having its registered office at Garrison Chapel, Castille Place, Valletta, VLT 1063, Malta;
Manoel Island	the divided portion of land at Manoel Island, limits of Gzira, comprised within the Emphyteutical Land as shown bordered in red on the plan Land Drawing letter 'L' letter 'D' one hundred and seventy four letter 'A' bar ninety nine (LD174A/99) attached to the Emphyteutical Deed;
MFSA	the Malta Financial Services Authority, incorporated in terms of the Malta Financial Services Authority Act (Cap. 330 of the laws of Malta);
Offer or Bond Offer	the invitation to subscribe for Bonds contained in the Prospectus;
Official List	the list prepared and published by the Malta Stock Exchange as its official list in accordance with the Malta Stock Exchange Bye-Laws;
Preferred Applicants	Existing Holders and Current Shareholders as at the Cut-Off Date, excluding their spouses and dependents;
Preferred Applicants Offer Period	the period between the 01 July 2016 and 18 July 2016 during which the Bonds are on offer to Preferred Applicants;
Public Offer Period	the period between the 04 July 2016 and 20 July 2016 (or such earlier date as may be determined by the Issuer) during which the Bonds are on offer to the general public;
Prospectus	collectively the Summary Note, the Registration Document and this Securities Note, all dated 28 June 2016, as such documents may be amended, updated, replaced and supplemented from time to time;
Prospectus Directive	Directive 2003/71/EC of the European Parliament and of the Council of 4 November 2003 on the prospectus to be published when securities are offered to the public or admitted to trading and amending Directive 2001/34/EC, as subsequently amended;

Redemption Date	27 July 2026;
Redemption Value	the nominal value of each Bond;
Registration Document	the registration document issued by the Issuer dated 28 June 2016, forming part of the Prospectus;
Securities Note	this document in its entirety;
Security Interest	the Initial Security Interest as may be varied or reduced from time to time in terms of clauses 8 and 9 of the Trust Instrument and any other security which may be held in trust for the Bondholders as beneficiaries under the terms of the Trust Instrument, including any undertaking, guarantee, mandate, pledge, title, transfer, grant, privilege or hypothec or the placing of property in possession or control of the Security Trustee with rights of retention and sale;
Security Trustee	CSB Trustees & Fiduciaries Limited having company registration number C 40390 and its registered office at Vincenti Buildings, 28/19 Strait Street, Valletta VLT 1432, Malta, licensed by the MFSA to act as trustee and provide general corporate fiduciary services, or any other duly authorised person as may be appointed to act as security trustee in terms of the Trust Instrument;
Summary Note	the summary note issued by the Issuer dated 28 June 2016, forming part of the Prospectus;
Tigné Point	the divided portion of land at Tigné Point, Sliema, comprised within the Emphteutical Land as shown bordered in red on the plan Land Drawing letter 'L' letter 'D' one hundred and seventy five letter 'A' bar ninety nine (LD175A/99) attached to the Emphyteutical Deed;
Trust Instrument	the agreement signed between the Issuer and the Security Trustee dated 24 June 2016 as better described in section 9.6 of this Securities Note.

2. RISK FACTORS

THE VALUE OF INVESTMENTS CAN GO UP OR DOWN AND PAST PERFORMANCE IS NOT NECESSARILY INDICATIVE OF FUTURE PERFORMANCE.

AN INVESTMENT IN THE BONDS INVOLVES CERTAIN RISKS INCLUDING THOSE DESCRIBED BELOW. PROSPECTIVE INVESTORS SHOULD CAREFULLY CONSIDER, WITH THEIR OWN INDEPENDENT FINANCIAL AND OTHER PROFESSIONAL ADVISERS, THE FOLLOWING RISK FACTORS AND OTHER INVESTMENT CONSIDERATIONS AS WELL AS ALL THE OTHER INFORMATION CONTAINED IN THE PROSPECTUS BEFORE DECIDING TO MAKE AN INVESTMENT IN THE BONDS. THE SEQUENCE IN WHICH THE RISKS BELOW ARE LISTED IS NOT INTENDED TO BE INDICATIVE OF ANY ORDER OF PRIORITY OR OF THE EXTENT OF THEIR CONSEQUENCES.

NEITHER THIS SECURITIES NOTE, NOR ANY OTHER PARTS OF THE PROSPECTUS OR ANY OTHER INFORMATION SUPPLIED IN CONNECTION WITH THE BONDS: (I) IS INTENDED TO PROVIDE THE BASIS OF ANY CREDIT OR OTHER EVALUATION OR (II) SHOULD BE CONSIDERED AS A RECOMMENDATION BY THE ISSUER OR AUTHORISED FINANCIAL INTERMEDIARIES THAT ANY RECIPIENT OF THIS SECURITIES NOTE OR ANY OTHER PART OF THE PROSPECTUS OR ANY OTHER INFORMATION SUPPLIED IN CONNECTION WITH THE PROSPECTUS OR ANY BONDS, SHOULD PURCHASE ANY BONDS.

ACCORDINGLY, PROSPECTIVE INVESTORS SHOULD MAKE THEIR OWN INDEPENDENT EVALUATION OF ALL RISK FACTORS, AND SHOULD CONSIDER ALL OTHER SECTIONS IN THIS DOCUMENT.

2.1 Risks relating to the Bonds

An investment in the Bonds involves certain risks including, but not limited to those described below:

Trading and Liquidity

There can be no assurance that an active secondary market for the Bonds will develop or, if it develops, that it will continue. Nor can there be any assurance that an investor will be able to re-sell his Bonds at or above the Bond Issue Price or at all. A public trading market having the desired characteristics of depth, liquidity and orderliness depends on a number of factors including the presence in the market place of willing buyers and sellers of the Issuer's Bonds at any given time, which presence is dependent upon the individual decisions of investors over which the Issuer has no control. Many other factors over which the Issuer has no control may affect the trading market for, and trading value of, the Bonds. These factors include the time remaining to the maturity of the Bonds, the outstanding amount of the Bonds and the level, direction and volatility of market interest rates generally. No prediction can be made about the effect which any future public offerings of the Issuer's securities or any takeover or merger activity involving the Issuer will have on the market price of the Bonds prevailing from time to time.

Interest Rate Risk

Investment in the Bonds involves the risk that subsequent changes in market interest rates may adversely affect the value of the Bonds.

Changes in Laws and Regulations

The Bonds are based on the requirements of the Listing Rules of the Listing Authority, the Companies Act and the Commission Regulation in effect as at the date of this Prospectus. No assurance can be given as to the impact of any possible judicial decision or change in law or administrative practice after the date of this Prospectus.

Amendments to the Bond Conditions

The Bond Conditions contain provisions in section 11.9 of this Securities Note for calling meetings of Bondholders in the event that the Issuer wishes to amend any of the Bond Conditions. These provisions permit defined majorities to bind all Bondholders including Bondholders who did not attend and vote at the relevant meeting and Bondholders who vote in a manner contrary to the majority.

2.2 Risks relating to the Security Interest

An investor in the Bonds will bear certain risks relating to the Security Interest according to the ranking of the relative Security Interest.

- i. Security interests protecting third party interests are identified in section 9.5 of this Securities Note; in respect of the special hypothec (forming part of the Initial Security Interest) enjoyed by the Bonds, third party interests have been registered which will rank in priority to Bondholders against the assets of the Issuer for so long as such security interests remain in effect, as explained further in sections 9.4 and 9.5 of this Securities Note.
- ii. By acquiring the Bonds, the Bondholder is considered to be bound by the terms of the Trust Instrument as if he had been a party to it. The Trust Instrument contains a number of provisions, which the investors ought to be aware of prior to acquiring the Bonds and, therefore, investors ought to read the description of the Trust Instrument contained in section 9.6 before acquiring the Bonds. For instance, in terms of the Trust Instrument:
 - a. The Security Trustee is not liable for any loss or expense attributable to any action taken or omitted to be taken by the Security Trustee, or any person appointed by the Security Trustee, unless the loss or expense is shown to have been caused by the gross negligence or misconduct of the Security Trustee or the person so appointed;
 - b. The Security Trustee is not bound to take any such steps or proceedings or take any other action to

enforce the security constituted by the Security Interest unless the Security Trustee shall have been indemnified to its satisfaction against all actions, proceedings, claims and demands to which it may thereby render itself liable and all costs, charges, damages and expenses which it may incur by so doing;

- c. the Security Trustee is not bound to declare the Bonds to have become immediately due and repayable in the case of an Event of Default, described in section 11.7 of this Securities Note, under the Bonds Conditions found in this Securities Note, unless requested to do so by a resolution passed by Bondholders holding not less than seventy-five per cent (75%) in nominal value of the Bonds then outstanding at a meeting of the Bondholders;
- d. the Bondholders are entitled to require the Security Trustee to convene a meeting of the participation of Bondholders; and
- e. the Security Trustee is entitled to accept reductions in and/or variations to the Security Interest.

3. PERSONS RESPONSIBLE

This document includes information given in compliance with the Listing Rules for the purpose of giving information with regard to the Issuer and the Bonds. All of the Directors, whose names appear under the heading "Persons Responsible" in section 3.1 of the Registration Document, accept responsibility for the information contained herein.

To the best of the knowledge and belief of the Directors, who have taken all reasonable care to ensure that such is the case, the information contained in this document is in accordance with the facts and does not omit anything likely to affect the import of such information. The Directors accept responsibility accordingly.

4. ADVISERS TO THE ISSUER, AUDITORS AND SECURITY TRUSTEE

4.1 Advisers to the Issuer

Legal Advisers

Mamo TCV Advocates Palazzo Pietro Stiges 103, Strait Street Valletta. VLT 1436 Malta

Sponsor

Charts Investment Management Service Limited Valletta Waterfront, Vault 17 Pinto Wharf Floriana FRN 1913 Malta

Manager and Registrar

Bank of Valletta p.l.c. BOV Centre Cannon Road St. Venera SVR 9030 Malta

4.2 Statutory Auditors and Financial Advisers

PricewaterhouseCoopers 78, Mill Street Qormi QRM 3101 Malta

4.3 Security Trustee

Registered office: Vincenti Buildings 28/19 Strait Street Valletta VLT 1432 Malta

Correspondence address: CSB Trustees & Fiduciaries Limited The Penthouse, Tower Business Centre Tower Street Swatar BKR 4013 Malta

5. ESSENTIAL INFORMATION

5.1 Interest of Natural and Legal Persons involved in the Offer

Bank of Valletta p.l.c. holds, directly and indirectly through its 50% shareholding in MSV Life p.l.c., a substantial shareholding (that is, in excess of 10%) in the Issuer and is also acting as Manager and Registrar. Lombard Bank Malta p.l.c. is a shareholder of the Issuer. Charts Investment Management Service Limited is acting as Sponsor and Mediterranean Bank plc is a sister company of the Sponsor. Each of the aforementioned entities is included in the list of Authorised Financial Intermediaries.

Each of the aforementioned entities has in place a 'Conflict of Interest Policy' and procedures and controls designed to identify, prevent or manage conflicts of interest. Where a conflict of interest is considered to arise, the subject entity is obliged by law to manage such conflict in the best interest of prospective investors, in line with the said internal policy, procedures and controls. Consequently, before proceeding to subscribing to any Bonds, a prospective investor shall be informed of the said conflict of interest.

Save for the above and any fees payable in connection with the Bond Issue to Charts Investment Management Service Limited as Sponsor and Bank of Valletta p.l.c. as Manager and Registrar, so far as the Issuer is aware, no person involved in the Offer has any other interest that is material to the Offer.

5.2 Reasons for the Offer and Use of Proceeds

The proceeds from the Bond Issue, net of expenses, are expected to amount to circa €49,000,000. However, the value of Bonds acquired by Existing Holders by virtue of the Bond Exchange Programme detailed in section 8.5 of this Securities Note will not be paid in cash. The net proceeds will be used by the Issuer for the following purposes, in the amounts and order of priority set out below:

(i) an amount of up to circa €40,832,900 of the net proceeds, less the value of the Bonds acquired through the Bond Exchange Programme, shall be used for the purpose of redeeming any Bonds 2016/18 remaining in issue as at 15 December 2016, being the first early date of redemption of the Bonds 2016/18. As at the date of the Prospectus the value of Bonds 2016/18 in issue stands at circa €40,832,900 (being the aggregate of

€31,702,900 EUR Bonds 2016/18 and £7,214,300 (equivalent to *circa* €9,130,000) GBP Bonds 2016/18). The amount of the principal still outstanding on the Bonds 2016/18 and any interest payable thereon will be dealt with as detailed in section 9.6.2.2, paragraph (i) of this Securities Note;

- (ii) the amount of circa €1,500,000 shall be used to obtain the release of the security interest provided by the Company to secure loan and overdraft facilities made available by HSBC Bank Malta p.l.c. to Solutions and Infrastructure Services Limited ("SIS"), a wholly owned subsidiary of the Issuer ("SIS Banking Facilities");
- (iii) the amount of circa €2,200,000 shall be advanced to Bank of Valletta p.l.c. as cash collateral in order for the aforementioned bank to cancel the security interest securing the obligations of the Company in terms of the euro/sterling cross currency interest rate swap instrument ("SWAP Instrument"). Upon termination of the SWAP Instrument, any residual amount from the aforesaid €2,200,000 will be used for general corporate funding purposes of the Group and/or to further reduce the corporate indebtedness of the Group;
- (iv) the amount of €4,500,000 shall be used to fund various infrastructural and restoration works at Tigné Point which are deemed essential for closing off the Tigné Point project.

In the event that the Bond Issue is not fully subscribed, the proceeds from the Bond Issue shall be applied in the manner and order of priority set out above. Any residual amounts required by the Issuer for the purposes of the uses specified in (i) and (ii) above which shall not have been raised through the Bond Issue shall be financed from the Group's general cash flow and/or bank financing.

All proceeds from the Bond Issue shall be held by the Security Trustee pending the fulfilment of the conditions indicated in section 9.6.2.1 of this Securities Note and in the event that any of the said conditions is not satisfied:

- (i) the Security Trustee shall return Bond proceeds received back to Applicants; and
- (ii) no Bonds 2016/18 shall be transferred in the Issuer's name in consideration for the issue of Bonds in terms of the Bond Exchange Programme detailed in section 8.5 of this Securities Note, and as such Existing Holders shall retain title of their respective holding of Bonds 2016/18.

Furthermore, following the satisfaction of the conditions set out in section 9.6.2.1 of this Securities Note, certain amounts shall be held by the Security Trustee and dealt with as specified in section 9.6.2.2 (paragraphs (ii) and (iii) respectively) of this Securities Note.

5.3 Expenses

Professional fees, costs related to publicity, advertising, printing, listing and registration, selling commission, as well as sponsor, manager and registrar fees and other miscellaneous expenses in connection with this Bond Issue, are estimated not to exceed €1,000,000 and shall be borne by the Issuer.

6. OFFER STATISTICS

Issuer:	MIDI p.l.c., a company registered in Malta with registration number C 15836;
Issue:	4% Secured Bonds due 2026;
ISIN:	MT0000421223;
Amount:	Fifty million euro (€50,000,000);
Form:	The Bonds will be issued in fully certificated and registered form, without interest coupons. If and for as long as the Bonds are admitted to listing on the Malta Stock Exchange, certificates will not be delivered to Bondholders in respect of the Bonds as each Bondholder will be registered in dematerialised form by an appropriate entry in the electronic register maintained on behalf of the Issuer at the CSD or as may be stipulated by the MSE Bye-Laws from time to time;
Denomination:	euro (€);
Minimum amount per subscription:	Minimum of two thousand euro (€2,000) and integral multiples of one hundred euro (€100);
Redemption Date:	27 July 2026;
Issue Price:	At par (€100 for each Bond);
Status of the Bonds:	The Bonds, as and when issued and allotted, shall constitute the general, direct, unconditional and secured obligations of the Issuer. They shall at all times rank <i>pari passu</i> without priority or preference among themselves. They shall rank subsequent to any other prior ranking indebtedness of the Company, if any, as detailed in sections 9.4 and 9.5 of this Securities Note;
Listing:	The Listing Authority has approved the Bonds for admissibility to listing. Application has been made to the Malta Stock Exchange for the Bonds to be listed and traded on its Official List;
Closing date for Applications:	18 July 2016 in the case of Preferred Applicants and 20 July 2016 (or such earlier date as may be determined by the Issuer in the event of over-subscription) in the case of the general public;
Preferred Applicants' Offer Period	the period between the 01 July 2016 and 18 July 2016 during which the Bonds are on offer to Preferred Applicants;
Public Offer Period:	The period between 04 July 2016 and 20 July 2016 (or such earlier date as may be determined by the Issuer in the event of over-subscription) during which the Bonds are on offer to the general public;
Interest:	4% per annum for each of the Bonds;
Yield:	The gross yield calculated on the basis of the Interest, the Issue Price and the Redemption Value of the Bonds at maturity is 4% for each of the Bonds;

Interest Payment Date(s):	Annually on 27 July as from 27 July 2017 (the first interest payment date);
Redemption Value:	At par (€100 for each Bond);
Manager / Registrar:	Bank of Valletta p.l.c.;
Sponsor:	Charts Investment Management Service Limited;
Security Trustee:	CSB Trustees & Fiduciaries Limited;
Notices:	Notices will be mailed to Bondholders at their registered addresses and shall be deemed to have been served at the expiration of twenty-four (24) hours after the letter containing the notice is posted, and in proving such service it shall be sufficient to prove that a prepaid letter containing such notice was properly addressed to such Bondholder at his/her registered address and posted;
Governing Law:	The Bonds are governed by and shall be construed in accordance with Maltese law;
Jurisdiction:	The Maltese Courts shall have exclusive jurisdiction to settle any disputes that may arise out of or in connection with the Bonds and accordingly any legal action or proceedings arising out of or in connection with the Bonds shall be brought exclusively before the Maltese Courts.

7. EXPECTED TIMETABLE

1	Application Forms mailed to Preferred Applicants	30 June 2016
2	Opening of Preferred Applicants Offer Period	01 July 2016
3	Application Forms available to the general public	04 July 2016
4	Closing of Preferred Applicants Offer Period	18 July 2016
5	Opening of Subscription Lists	18 July 2016
6	Closing of Subscription Lists	20 July 2016
7	Commencement of interest	27 July 2016
8	Announcement of basis of acceptance	27 July 2016
9	Refunds of unallocated monies	03 August 2016
10	Admission to listing on the MSE and dispatch of allotment advices	03 August 2016
11	Commencement of trading on the MSE	04 August 2016

The Issuer reserves the right to close Subscriptions Lists before 20 July 2016 in the event of over-subscription, in which case the events set out in steps 8 to 9 above shall be brought forward, although the number of working days between the respective events shall not also be altered.

8. DETAILS OF THE OFFER AND ADMISSION TO TRADING

8.1 The Bond Issue

The Issuer is issuing 4% Secured Bonds due 2026 for an aggregate principal amount of €50,000,000. The Bonds, having a nominal value of €100 each, will be issued at par and shall bear interest at the rate of 4% per annum payable annually in arrears on 27 July of each year, the first such payment to be made on 27 July 2017. The Bonds, unless previously purchased and cancelled, will be redeemed on 27 July 2026.

The Bond Issue is not underwritten. In the event that the Bond Issue is not fully subscribed, the Issuer will proceed with the listing of the amount of Bonds subscribed for and effect cancellation of the Bonds 2016/18 held by Existing Holders electing to acquire Bonds through the Bond Exchange Programme (as better described in section 8.5 of this Securities Note).

8.2 Distribution and Offer Period

The Bonds are open for subscription to all categories of investors, which may be broadly split as follows:

- (i) The Issuer has reserved an aggregate amount of Bonds equivalent to the amount of Bonds 2016/18 of circa €40,832,900, subject to any cash top-up as applicable (any Existing Holder whose holding in Bonds 2016/18 is less than the equivalent amount of €2,000 is required to pay the difference up to the minimum subscription amount of €2,000), for subscription by Existing Holders through the Bond Exchange Programme during the Preferred Applicants Offer Period;
- (ii) The Issuer has reserved an aggregate amount of Bonds amounting to €2,000,000 for subscription by Current Shareholders during the Preferred Applicants Offer Period;
- (iii) The remaining balance of *circa* €7,167,100 in Bonds shall be made available for subscription by: (a) Existing Holders in respect of any number of additional Bonds applied for, other than through the Bond Exchange Programme, exceeding in value the nominal amount of EUR Bonds 2016/18 and/or the equivalent euro value of the nominal amount of GBP Bonds 2016/18 held by them as at the Cut-Off Date, during the Preferred Applicants Offer Period; and (b) the general public during the Public Offer Period.

In the event that the aggregate amount of *circa* €42,832,900 reserved for the Preferred Applicants (detailed in clauses (i) and (ii) above) is not fully taken up, such unutilised portion/s shall also become available for allocation to subscribers in terms of clause (iii) above.

In the event that subscriptions exceed the reserved portion referred to in clause (ii) above of €2,000,000, the unsatisfied excess amounts of such Applications will automatically participate in the amount of Bonds available in clause (iii) above.

The minimum subscription of Bonds must be €2,000 and any subscription thereafter must be in multiples of €100. Applications for subscription to the Bonds may be made through any of the Authorised Financial Intermediaries listed in Annex II of this Securities Note.

8.3 Allocation Policy

The Issuer shall allocate the Bonds on the basis of the following policy and order of priority:

i. An aggregate amount of Bonds equivalent to the amount of Bonds 2016/18 (*circa* €40,832,900) shall be allocated to Existing Holders applying for Bonds by way of the Bond Exchange Programme, subject to a minimum amount of €2,000 and in multiples of €100.

- ii. An aggregate amount of €2,000,000 has been reserved for subscription by Current Shareholders and shall be allocated in accordance with the allocation policy as determined by the Issuer and Registrar;
- iii. Following the allocation referred to in paragraphs (i) and (ii) hereof, the remaining Bonds shall be allocated to:
 - (a) Existing Holders having applied for Bonds in excess of their respective holding in the Bonds 2016/18;
 - (b) Current Shareholders with respect to such unsatisfied excess amount that may result from an oversubscription in the reserved portion of €2,000,000; and
 - (c) Applications submitted by the general public,

without priority or preference and in accordance with the allocation policy as determined by the Issuer and Registrar.

The Issuer will announce the allocation policy for the allotment of the Bonds through a company announcement within five (5) Business Days of the closing of the Public Offer Period.

8.4 Allotment Results

It is expected that an allotment advice to Applicants will be dispatched within five (5) Business Days of the announcement of the allocation policy. The registration advice and other documents and any monies returnable to Applicants may be retained pending clearance of the remittance and any verification of identity as required by the Prevention of Money Laundering Act (Cap. 373 of the laws of Malta), and regulations made thereunder. Such monies will not bear interest while retained as aforesaid.

Listing shall be effected subject to the fulfilment of the conditions set out in section 9.6.2 of the Securities Note. Dealing shall commence upon admission to trading of the Bonds by the MSE, and subsequent to the above mentioned notification.

8.5 Bond Exchange Programme

Existing Holders of EUR Bonds 2016/18 applying for Bonds may elect to settle all or part of the amount due on the Bonds applied for by the transfer to the Issuer of EUR Bonds 2016/18 at par value, subject to a minimum holding of €2,000 in Bonds. Accordingly, any Existing Holder whose holding in EUR Bonds 2016/18 is less than €2,000 shall be required to pay the difference together with the submission of their Application Form 'A'.

Existing Holders of GBP Bonds 2016/18 applying for Bonds may elect to settle all or part of the amount due on the Bonds applied for by the transfer to the Issuer of GBP Bonds 2016/18. For such purpose, the par value of each holding in GBP Bonds 2016/18 shall be converted to the equivalent value in euro at the exchange rate of \le 1: \le 0.834, rounded up to the nearest \le 100. In determining the aforesaid exchange rate, the Issuer was guided by the reference exchange rate published by the European Central Bank on 27 June 2016 at 15:30 hours'. Subscriptions shall be subject to a minimum holding of \le 2,000 in Bonds. Accordingly, any Existing Holder whose holding in GBP Bonds 2016/18 is less than the equivalent amount of \le 2,000 shall be required to pay the difference in Euro together with the submission of their Application Form 'B'.

Holders of EUR Bonds 2016/18 and GBP Bonds 2016/18 shall use Application Forms 'A' and 'B' respectively, to be mailed directly by the Issuer and shall be required to submit same to Authorised Financial Intermediaries together with cleared funds, if applicable (as described below), during the Preferred Applicants Offer Period.

By virtue of the submission of the duly completed and signed Application Form, Existing Holders shall indicate their agreement to settle the consideration for the Bonds by surrendering in the Issuer's favour for cancellation all or part of the Bonds 2016/18 together with the payment of such additional amount in cash as may be required, and such Application Form shall be deemed:

- (i) to confirm that all or part (as the case may be) of the said Bonds 2016/18 as indicated in the Application Form are to be transferred in the Issuer's favour for cancellation; and
- (ii) as an irrevocable mandate to the Issuer to engage the services of such brokers or intermediaries as may be necessary to fully and effectively carry out all procedures necessary and to fully and effectively vest title in the appropriate number of Bonds in the Existing Holder.

Any Existing Holder wishing to surrender and cancel all or part of their Bonds 2016/18 in exchange of the Bonds shall only be entitled to do so during the Preferred Applicants Offer Period.

8.6 Refunds

If any Application is not accepted, or if any Application is accepted for fewer Bonds than those applied for, the Application monies or the balance of the amount paid on Application will be returned by the Issuer, without interest, by direct credit into the Applicant's bank account as indicated by the Applicant on the relevant Application Form within five (5) Business Days from the date of announcement of acceptance of Bonds (refer to section 7 of this Securities Note).

Neither the Issuer nor the Registrar will be responsible for any charges, loss or delays in transmission of the refunds.

In this regard, any monies returnable to Applicants may be retained pending clearance of the remittance or surrender of the Bonds 2016/18, as the case may be, and any verification of identity as required by the Prevention of Money Laundering Act (Cap. 373 of the laws of Malta) and regulations made thereunder. Such monies will not bear interest while retained as aforesaid.

8.7 Admission to Trading

The Listing Authority has authorised the Bonds as admissible to Listing pursuant to the Listing Rules by virtue of a letter dated 28 June 2016.

Application has been made to the Malta Stock Exchange for the Bonds being issued pursuant to the Prospectus to be listed and traded on the Official List of the Malta Stock Exchange.

The Bonds are expected to be admitted to the Malta Stock Exchange with effect from 03 August 2016, and trading is expected to commence on 04 August 2016.

9. INFORMATION CONCERNING THE BONDS

9.1 Description of the Bonds

Once issued, the Bonds shall constitute the debt obligations of the Issuer that bind the Issuer to pay to the Bondholders interest on each Interest Payment Date and the nominal value of the Bonds on the Redemption Date. The Bonds shall be issued at a nominal value of €100 for each Bond. The Bonds have been created in terms of the Companies Act. The issue of the Bonds is being made in accordance with the requirements of the Listing Rules, the Companies Act, and the Commission Regulation.

The Bonds will be issued in dematerialised form, and following admission of the Bonds to the Official List of the Malta Stock Exchange, the Bonds shall be held in book-entry form by the CSD of the Malta Stock Exchange.

Subject to admission to listing of the Bonds to the Official List of the MSE, the Bonds are expected to be assigned ISIN code MT0000421223.

9.2 Currency of the Bond Issue

The currency of the Bonds is euro (€).

9.3 Rights attached to the Bonds

There are no special rights attached to the Bonds other than the right of the Bondholders to:

- (i) the payment of capital;
- (ii) the payment of interest;
- (iii) the benefit of the Security Interest through the Security Trustee;
- (iv) attend, participate in and vote at meetings of Bondholders in accordance with section 11 of this Securities Note ("Terms and Conditions of the Bonds"); and
- (v) enjoy all such other rights attached to the Bonds emanating from this Prospectus.

9.4 Summary of Existing Registered Privileges and Hypothecs

A summary of existing registered privileges and hypothecs resulting from searches carried out up to the 06 May 2016 is attached to this Securities Note as Annex III.

9.5 The Security Interest

The Bonds shall be secured by the Security Interest held by the Security Trustee in favour of the Bondholders as beneficiaries of the Trust Instrument. Details of the Trust Instrument are contained in section 9.6 of this Securities Note.

Prior to the issue and allotment of the Bonds, the Bonds shall be secured by the Initial Security Interest. The Initial Security Interest may be varied or reduced from time to time, as provided for in clauses 8 and 9 of the Trust Instrument, so however that the value of the security forming part of the Security Interest from time to time shall not be less than the aggregate nominal amount of the Bonds and the interest payable thereon. Further detail on this matter is found in section 9.6 below. In the event that the Initial Security Interest be varied as aforesaid, the ranking of the Security Interest may not be the same as the ranking of the Initial Security Interest indicated in Annex III. Moreover, the Security Trustee, if it is deemed to be in the interest of Bondholders, may provide its consent for security to be granted by the Issuer which shall rank prior to the Initial Security Interest in terms of and under the conditions set out in clauses 8 and 9 of the Trust Instrument, described in section 9.6.2 of this Securities Note. In such cases, the Security Trustee shall, in giving his consent, act for the benefit of Bondholders, as he is obliged to do in terms of law (article 3.1 of the Trusts and Trustees Act (Cap. 331 of the laws of Malta).

The **Initial Security Interest** shall consist of the following:

- (i) the special hypothec to be constituted by the Issuer in favour of the Security Trustee by virtue of a deed of hypothec for the amount of fifty million euro (€50,000,000) and interest and costs thereon, over the immovable property detailed below under the heading "Summary of Security in favour of the Security Trustee" (sub-headings: "Commercial premises"; "Car parking spaces"; "Storage rooms"; and "Properties earmarked for development"). This special hypothec shall rank as indicated below (see also Annex III).
- (ii) the pledge of 11,699,999 shares in T14 Investments Limited (C 63982), detailed below under the heading "Summary of Security in favour of the Security Trustee" (sub-heading: "Shareholding in T14 Investments Ltd"), to be constituted by the Company in favour of the Security Trustee by virtue of the pledge agreement for the principal amount of the Bonds or any amount outstanding and any interest thereon. These shares have not been previously pledged.

MIDI p.l.c.

Summary of Security in favour of the Security Trustee

Summary of Security in favour of the Security mustee	€'000
Commercial premises:	
Q2 Block - ground floor	3,237
T4P, T7P and T9P Blocks - 11 outlets	11,885
T12 Clubhouse - 2 catering outlets, health & leisure facility and car parking spaces	6,518
Fort Tigné	3,876
	25,516
Car parking spaces:	
T11 Section, T8/T9/T10 Section (level -4) - 519 spaces	12,975
T1 Section - 132 spaces	3,300
'Q' car park (level -2) - 62 spaces	1,550
Car park behind T2 Section - 35 spaces	875
T15/T16 Section - 39 spaces	975
	19,675
Storage rooms:	
Total area of <i>circa</i> 3,216m ²	1,929
1000 0100 0100 0100	
Properties earmarked for development:	
T20 site	636
Shareholding in T14 Investments Ltd:	
Value of 11,699,999 shares in T14 Investments Ltd	11,700
Less Government's privilege on T14 Site	(6,037)
	5,663
Total	53,419

^{*} Note on item "Shareholding in T14 Investments Ltd": In terms of a public deed in the records of Notary Dr Pierre Attard of 12 July 2014 between inter alia the Issuer and Mid Knight Holdings Limited ("MKH"), the Issuer transferred the site for the development of the T14 Building to MKH. The said site was burdened by a special hypothec in favour of the GOM, as security for the payment of the premium, and this is better described in Annex III. On the said deed, the Issuer remained responsible to pay the amount of €6,037,155.24 and to obtain the cancellation of the registration of such hypothec. MKH is entitled in terms of the said deed to make payments of capital or dividends due to the MIDI Group in a pledged account in favour of MKH as security pending the cancellation of the said special hypothec. The value of this special hypothec has been taken into consideration when establishing the value of the Initial Security Interest to be given to the Security Trustee.

In terms of article 1995 of the Civil Code (Cap. 16 of the laws of Malta), the property of a debtor is the common guarantee of his creditors, all of whom have an equal right over such property, unless there exist between them "lawful causes of preference." Privileges and hypothecs are "lawful causes of preference".

A privilege is a right of preference which the nature of a debt confers upon a creditor over the other creditors, including hypothecary creditors (article 1999 of the Civil Code). A debt due to a pledgee is considered a privileged debt over the thing which is held as a pledge (article 2009(a) of the Civil Code). Securities, including shares, may be pledged by their holder in favour of any person as security for any obligation (article 122(1) of the Companies Act). In the event of default under the agreement constituting the pledge, and upon giving notice by judicial act to the pledgor and the company whose shares are pledged, the pledgee is entitled to dispose of the securities which are pledged in his favour or appropriate and acquire the securities himself in settlement of the debt due or part thereof. This is without prejudice to the right of the pledgee to apply for judicial sale of the securities and notwithstanding the provisions of the Civil Code or of the memorandum and articles of the company whose shares are pledged (Article 122(6) of the Companies Act).

A hypothec is a right created over the property of a debtor or of a third party for the benefit of the creditor, as security for the fulfilment of an obligation (article 2011 of the Civil Code); a special hypothec is a hypothec which affects particular immovables (article 2012 of the Civil Code).

Debts which are secured by privileges rank according to the order set out by law (articles 2089 to 2091 of the Civil Code), whilst debts which are secured by hypothecs rank according to the date of registration(article 2092 of the Civil Code).

The pledge of 11,699,999 shares in T14 Investments Limited therefore would constitute a privilege in favour of the Security Trustee, on behalf of the Bondholders, which in terms of article 2089 of the Civil Code means that the principal amount of the Bonds still outstanding together with accrued interest ranks prior to other debts over the shares so pledged. Moreover the Security Trustee, acting on behalf of the Bondholders, may exercise any of the remedies specified at law in order to enforce the pledge over the said shares.

The special hypothec over the immovable property indicated above would secure the principal amount of the Bonds still outstanding together with accrued interest however in the event of enforcement of the special hypothec, the Security Trustee, on behalf of the Bondholders, will rank after the persons and/or entities indicated below (see also Annex III), and in preference to any future creditors, save in situations where the Security Trustee consents to security to be granted by the Issuer ranking prior to the Initial Security Interest as noted above.

Following the publication of the relevant deeds of postponement, waiver, reduction and/or cancellation of the general hypothecs and/or special hypothecs described in Annex III to the Securities Note and identified therein as pending postponement, waiver, reduction or cancellation, as applicable; as well as the publication of the relative deed of hypothec and the execution of the pledge agreement constituting the Initial Security Interest, the Initial Security Interest shall **rank** as follows:

The Initial Security Interest which consists of the pledge of 11,699,999 shares in T14 Investments will be first ranking with respect to the said shares.

The Initial Security Interest over the immovable property indicated above will rank after the following:

- (a) the security interest which the GOM retained in its capacity as the holder of the *directum dominium* over the Emphyteutical Land, as security for the payment of ground rent which security interest follows the transfer of land; and
- (b) any security interest which may arise by operation of law.

The Initial Security Interest over the immovable property will also rank after the general hypothecs granted by the Issuer in favour of purchasers of property on the deeds of sale, over all the Issuer's property present and future in warranty of peaceful possession. On the final deeds of sale the purchasers specifically renounced to their rights under section 2016 of the Civil Code to cause to be registered, as a further security, a special hypothec on the Company's property.

Moreover, the security interests granted by the Company to secure the SIS Banking Facilities (vide items 12 and 13 in Annex III) and the obligations of the Company in terms of the SWAP Instrument (vide items 2, 4 and 11 in Annex III) shall also rank prior to the Initial Security Interest over the immovable property until such time as the events described in section 9.6.2.2 (ii) and (iii) of this Securities Note occur.

It should also be noted that in terms of the Trust Instrument the Security Trustee may retain and pay to itself out of any monies or the proceeds of any investments in its hands upon the trusts of the Trust Instrument all sums owing to it in respect of remuneration costs, charges, expenses or interest or by virtue of any indemnity from the Company to which it is entitled under the Trust Instrument or by law or by virtue of any release or indemnity granted to it and all such sums as aforesaid shall be so retained and paid in priority to the claims of the Bondholders and shall constitute an additional charge upon the property charged with the Security Interest.

Pursuant to the provisions of the Trust Instrument, the Security Trustee shall retain all proceeds from the Bonds until such time as the Initial Security Interest shall have been duly constituted in favour of the Security Trustee. No Bonds shall be issued and allotted until the Initial Security Interest has been duly constituted and the Malta Stock Exchange admits the Bonds to trading as listed instruments.

9.6 The Trust Instrument

The following does not purport to constitute an exhaustive summary of the Trust Instrument. This section 9.6 is simply intended to give an overview of the more salient provisions of the Trust Instrument. For more information on the provisions of the Trust Instrument, please refer to the Trust Instrument which is available for inspection as indicated in section 18 of the Registration Document.

Terms in this section 9.6 have the meaning assigned to them in the Trust Instrument.

9.6.1 General provisions

The trust is constituted in terms of Article 2095E of the Civil Code (Cap. 12 of the laws of Malta), and is to be treated as constituted in the context of a commercial transaction in terms and for the purposes of the Trusts and Trustees Act (Cap. 331 of the laws of Malta) (Clauses 3.2 and 3.3).

9.6.2 Safeguards for the Applicants and Existing Holders

9.6.2.1 Conditions

The net proceeds from the issue of the Bonds shall be held by the Security Trustee pending the fulfillment by the Issuer of the following conditions (Clause 4.3):

- (i) the publication of the deed/s pertaining to the postponements, waivers, reductions or cancellations of the general hypothecs and/or special hypothecs described in Annex III to the Securities Note and identified therein as pending postponement, waiver, reduction or cancellation, as applicable;
- (ii) the publication of the deed of hypothec forming part of the Initial Security Interest;
- (iii) the execution of the pledge agreement forming part of the Initial Security Interest; and
- (iv) the admission of the Bonds to the Official List of the MSE.

9.6.2.2 Transfer of the net proceeds of the Bonds to the Issuer

Upon the satisfaction of the conditions indicated in section 9.6.2.1 above, the Security Trustee shall transfer the net proceeds from the issue of the Bonds to the Issuer less:

- (i) the Reserved Amount (if any), which shall be transferred by the Security Trustee to the Reserve Account referred to in the agreement entered into between the Company and HSBC Bank Malta p.l.c. dated 23 December 2010, where the Reserved Amount is the difference, if any, between:
 - (a) the amount of principal still outstanding on the Bonds 2016/18 and any interest payable thereon until the date of maturity after taking into account the Existing Holders who do not participate in the Bond Exchange Programme described in section 8.5 of this Securities Note, or who do not participate fully in the said Bond Exchange Programme; and

- (b) the value of any assets held in trust by HSBC Bank Malta p.l.c. in virtue of an agreement entered into between the Company and HSBC Bank Malta p.l.c. dated 23 December 2010, which assets are held in trust in favour of the Existing Holders.
- (ii) the amount of circa €1,500,000, which shall be transferred by the Security Trustee to the Issuer or to any person who the Issuer may delegate the Security Trustee to transfer the said amount to, for the purposes of and upon the release of the security interest described in items 12 and 13 of Annex III, registered in favour of HSBC Bank Malta p.l.c., granted by the Company to secure the SIS Banking Facilities; and
- (iii) the amount of circa €2,200,000, which shall be transferred by the Security Trustee to the Issuer or to any person who the Issuer may delegate the Security Trustee to transfer the said amount to, for the purposes of and upon the cancellation of the security interest described in item 2, 4 and 11 of Annex III, registered in favour of Bank of Valletta p.l.c., securing the obligations of the Company in terms of the SWAP Instrument.

9.6.3 Variation or reduction of the Security Interest

The Trust Instrument makes provision for the following situations:

- (i) In the event that the Bonds are purchased and cancelled by the Issuer, the value of the Security Interest is to be reduced by an amount equivalent to the principal amount of the Bonds so purchased and cancelled (Clause 7.4).
- (ii) The Company has the right to have a part or parts of the property being secured by the Security Interest to be released from the effects of the Security Interest and/or to have the relative Security Interest waived or postponed, without substituting other property or money, provided that the property remaining immediately after such release or waiver shall have an aggregate value as shown by a valuation of not less than the aggregate nominal amount of the Bonds and the interest payable thereon for a period of one (1) year. The Trust Instrument defines 'valuation' as a valuation made by such professional valuer as may be nominated or approved by the Security Trustee ("Valuer") on the basis of sale in the open market between a willing buyer and a willing seller or on such other basis as the Security Trustee may approve, in the case of immovable property, and as a valuation made by a Valuer on the basis of applicable international standards, or such other basis as the Security Trustee may approve, in the case of movable property. The Issuer and the Security Trustee have the right to require a valuation at any time at the Issuer's expense, and may require the release, waiver or postponement of the Security Interest as aforesaid within twelve (12) months of the date of the valuation. (Clause 8)
- (iii) The Company has the right to have all or any part of the property secured by the Security Interest to be released from the effects of the relative Security Interest and/or of having the relative Security Interest waived or postponed upon alternative security being provided in favour of and to the satisfaction of the Security Trustee. In such cases, the Valuer would have to certify that following substitution the aggregate value of the property secured by the security Interest shall be not less than the aggregate nominal amount of the Bonds and the interest payable thereon for a period of one (1) year. In this case, the Security Trustee may, subject to agreement with the Company, instead of requiring a valuation, treat the value of the property as being its proposed sale price. (Clause 9)

Furthermore, upon purchase and cancellation of all the Bonds or payment on redemption or otherwise of the principal amount of the Bonds, payment of all interest thereunder and reimbursement of all expenses incurred by, and payment of remuneration due to the Security Trustee under the Trust Instrument, the Security Interest shall be released, and the hypothecs, pledges or other security constituting the security Interest shall be cancelled.

9.6.4 Covenants and representations and warranties

The Company has covenanted various obligations with the Security Trustee which apply for the duration of the continuance of the Security Interest (Clause 11). These covenants may be enforced by the Security Trustee (Clause 15.1). Moreover, the Company has made a number of representations and warranties to the Security Trustee, that relies on such representations and warranties (Clause 13).

9.6.5 Functions and powers of the Security Trustee

The Trust Instrument makes provision for various powers of the Security Trustee. Most notably, the Security Trustee may, by notice in writing to the Issuer, declare the Bonds to have become immediately due and repayable in the case of an Event of Default under the Bond Conditions (section 11.7 of this Securities Note) and may take any proceedings against the Company as it may deem fit including for the enforcement of the Security Interest if the Company fails to pay the principal amount as and when the Bonds are due to be redeemed and failure continues for 30 days after written notice is given to the Company or at any time after the Bonds have become immediately due and repayable (Clause 12). The functions and powers of the Security Trustee are dealt with in Clause 15 of the Trust Instrument, which also provides that generally the Security Trustee shall not be liable for any error of judgement committed in good faith unless it is proved that it was grossly negligent in ascertaining the pertinent facts and the Security Trustee, its officers, employees and agents are entitled to be indemnified out of the property subject to the Security Interest so far as may be lawful in respect of all liabilities incurred in the execution of the trusts of the Trust Instrument.

The Security Trustee may pay to itself out of the trust fund all sums owing to it in respect of remuneration, costs, charge, expenses or interest, or by virtue of any indemnity from the Company to which it is entitled under the Trust Instrument or by law or by virtue of any release of indemnity granted to it, out of any monies or the proceeds of any investments in its hands upon the trusts of the Trust Instrument. All such sums shall be retained and paid in priority to the claims of the Bondholders and shall constitute an additional charge upon the property charged with the Security Interest. (Clause 18.3)

9.6.6 Resignation and removal of Security Trustee

The Security Trustee may resign as security trustee by giving not less than three (3) months' notice in writing to the Company. The Company may remove the Security Trustee by giving him at least one (1) months' notice.

In the event of the Security Trustee giving notice or being removed, the Company undertakes to procure a new trustee to be appointed ("the Successor Trustee"). The retirement or removal shall not become effective until such time as a Successor Trustee, is appointed and the Security Interest is transferred to the Successor Trustee. Upon appointment of the Successor Trustee, references contained in the Trust Instrument and this Prospectus to the Security Trustee shall be deemed to refer to the Successor Trustee.

The terms and conditions of the Trust Instrument shall, upon subscription or purchase of any Bonds, be binding on such subscriber or purchaser as a beneficiary under the trust as if the Bondholder had been a party to the Trust Instrument and as if the Trust Instrument contained covenants on the part of each Bondholder to observe and be bound by all the provisions hereof, and the Security Trustee is authorised and required to do the things required of it by the Trust Instrument.

9.7 Registration, Form, Denomination and Title

- 1. Certificates will not be delivered to Bondholders in respect of the Bonds in virtue of the fact that the entitlement to Bonds will be represented in an uncertificated form by the appropriate entry in the electronic register maintained on behalf of the Issuer by the CSD. There will be entered in such electronic register the names, addresses, identity card numbers (in the case of natural persons), registration numbers (in the case of companies) and MSE account numbers of the Bondholders and particulars of the Bonds held by them respectively, and the Bondholders shall have, at all reasonable times during business hours, access to the register of bondholders held at the CSD for the purpose of inspecting information held on their respective account.
- 2. The CSD will issue, upon a request by a Bondholder, a statement of holdings to such Bondholder evidencing his/her/its entitlement to Bonds held in the register kept by the CSD.
- 3. The Bonds will be issued in fully registered form, without interest coupons, in denominations of any integral multiple of €100, provided that on subscription the Bonds will be issued for a minimum of €2,000 per individual Bondholder. Authorised Financial Intermediaries subscribing to the Bonds through nominee accounts for and on behalf of clients shall apply the minimum subscription amount of €2,000 to each underlying client.

9.8 Interest

The Bonds shall accrue interest at the rate of 4% per annum, payable annually on the Interest Payment Date. Interest shall accrue as from 27 July 2016. The first Interest Payment Date shall be 27 July 2017 (covering the period 27 July 2016 to 26 July 2017). Further information is provided in section 11.4 of this Securities Note.

9.9 Transferability of the Bonds

The Bonds are freely transferable and, once admitted to the Official List of the MSE, shall be transferable only in whole in multiples of €100 in accordance with the rules and regulations of the MSE applicable from time to time. Further information on the transferability of the Bonds is provided in section 11.8 of this Securities Note.

9.10 Maturity and Redemption

The Bonds shall become due for redemption on 27 July 2026. Redemption of the Bonds shall be made at the nominal value of the Bonds. In addition, the Issuer reserves the right to purchase and cancel Bonds from the market at any time after issue.

9.11 Yield

The gross yield calculated on the basis of the Interest, the Bond Issue Price and the Redemption Value of the Bonds at Redemption Date is 4%.

9.12 Authorisations and Approval

The Board of Directors of the Issuer authorised the Bond Issue and the publication of the Prospectus pursuant to a board of directors' resolution passed on 26 April 2016.

9.13 Expected Issue Date of the Bonds

The expected Issue Date of the Bonds is 03 August 2016.

10. TAXATION

10.1 General

Investors and prospective investors are urged to seek professional advice as regards both Maltese and any foreign tax legislation which may be applicable to them in respect of the Bonds, including their acquisition, holding and disposal as well as any income/gains derived therefrom or made on their disposal. The following is a summary of the anticipated tax treatment applicable to Bondholders in so far as taxation in Malta is concerned. This information does not constitute legal or tax advice and does not purport to be exhaustive.

The information below is based on an interpretation of tax law and practice relative to the applicable legislation, as known to the Issuer at the date of the Prospectus, in respect of a subject on which no official guidelines exist. Investors are reminded that tax law and practice and their interpretation as well as the levels of tax on the subject matter referred to in the preceding paragraph, may change from time to time.

This information is being given solely for the general information of investors. The precise implications for investors will depend, among other things, on their particular circumstances and on the classification of the Bonds from a Maltese tax perspective, and professional advice in this respect should be sought accordingly.

10.2 Malta Tax on Interest

Since interest is payable in respect of a Bond which is the subject of a public issue, unless the Issuer is otherwise instructed by a Bondholder or if the Bondholder does not fall within the definition of "recipient" in terms of article 41(c) of the Income Tax Act (Cap. 123 of the laws of Malta), interest shall be paid to such person net of a final withholding tax, currently at the rate of 15% of the gross amount of the interest, pursuant to article 33 of the Income Tax Act. Bondholders who do not fall within the definition of a "recipient" do not qualify for the said rate and should seek advice on the taxation of such income as special rules may apply.

This withholding tax is considered as a final tax and a Maltese resident individual Bondholder need not declare the interest so received in his income tax return. No person shall be charged to further tax in respect of such income. Where the Bondholder is a Maltese resident individual, he is still entitled to declare the gross interest in the tax return, however the tax so deducted will not be available as a credit against the individual's tax liability or for a refund as the case may be.

In the case of a valid election made by an eligible Bondholder resident in Malta to receive the interest due without the deduction of final tax, interest will be paid gross and such person will be obliged to declare the interest so received in his income tax return and be subject to tax on it at the standard rates applicable to that person at that time. Additionally in this latter case the Issuer will advise the Inland Revenue on an annual basis in respect of all interest paid gross and of the identity of all such recipients unless the beneficiary is a non-resident of Malta. Any such election made by a resident Bondholder at the time of subscription may be subsequently changed by giving notice in writing to the Issuer. Such election or revocation will be effective within the time limit set out in the Income Tax Act.

In terms of article 12(1)(c) of the Income Tax Act, Bondholders who are not resident in Malta satisfying the applicable conditions set out in the Income Tax Act are not taxable in Malta on the interest received and will receive interest gross, subject to the requisite declaration/evidence being provided to the Issuer in terms of law.

10.3 Exchange of Information

Non-residents of Malta should note that payment of interest to individuals and certain residual entities residing in another EU Member State is reported on an annual basis to the Director General Inland Revenue, Malta, who will in turn exchange the information with the competent tax authority of the Member State where the recipient of interest is resident. This exchange of information takes place in terms of the Council Directive 2014/107/EU of 9 December 2014 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation.

10.4 Maltese Taxation on Capital Gains on Transfer of the Bonds

On the assumption that the Bonds would not fall within the definition of "securities" in terms of article 5(1)(b) of the Income Tax Act, that is, "shares and stocks and such like instrument that participate in any way in the profits of the company and whose return is not limited to a fixed rate of return", no tax on capital gains is chargeable in respect of transfer of the Bonds.

10.5 Duty on Documents and Transfers

In terms of article 50 of the Financial Markets Act (Cap. 345 of the laws of Malta) as the Bonds constitute financial instruments of a company quoted on a regulated market Exchange, as is the MSE, redemptions and transfers of the Bonds are exempt from Maltese duty.

INVESTORS AND PROSPECTIVE INVESTORS ARE URGED TO SEEK PROFESSIONAL ADVICE AS REGARDS BOTH MALTESE AND ANY FOREIGN TAX LEGISLATION APPLICABLE TO THE ACQUISITION, HOLDING AND DISPOSAL OF BONDS AS WELL AS INTEREST PAYMENTS MADE BY THE ISSUER. THE ABOVE IS A SUMMARY OF THE ANTICIPATED TAX TREATMENT APPLICABLE TO THE BONDS AND TO BONDHOLDERS. THIS INFORMATION. WHICH DOES NOT CONSTITUTE LEGAL OR TAX ADVICE. REFERS ONLY TO BONDHOLDERS WHO DO NOT DEAL IN SECURITIES IN THE COURSE OF THEIR NORMAL TRADING ACTIVITY.

11. TERMS AND CONDITIONS OF THE BONDS

The following are the terms and conditions, which will be applicable to the Bonds (the "**Bond Conditions**"). Each Bondholder and any person claiming through or under a Bondholder is deemed to have notice and knowledge of, and is bound by, these Bond Conditions.

The issue of these Bonds is being made subject to the provisions of these Bond Conditions and of the Trust Instrument. A Bondholder is deemed to have invested in the Bonds only after having received, read and understood the contents of these Bond Conditions, the Prospectus and the Trust Instrument and therefore only after having full knowledge of the information contained in these Bond Conditions, the Prospectus and the Trust Instrument and is accordingly deemed to have accepted all the terms and conditions set out in the Bond Conditions, the Prospectus and the Trust Instrument.

11.1 General

Each bond forms part of a duly authorised issue of 4% Secured Bonds due 2026 having a nominal value of €100 each (the "**Bonds**") of MIDI p.l.c. (the "**Issuer**") for an aggregate principal amount of €50,000,000 (save what is provided in section 11.10 of this Securities Note). The Bonds issued by the Issuer shall be issued in euro (€).

11.2 Form, Denomination, Title and Pricing

- 1. The Bonds will be issued without interest coupons in denominations of integral multiples of €100 provided that on subscription they will be issued for a minimum of €2,000.
- 2. On subscription, the Bonds are being offered at par, that is at €100 per Bond.
- 3. The Bonds will be issued in dematerialised form and shall accordingly be evidenced by a book-entry in the register of Bondholders held by the CSD of the Malta Stock Exchange or as may be stipulated by the MSE Bye-Laws from time to time.
- 4. The nominal value of each Bond is being established to facilitate trading therein. The bonds, and the transfer thereof, shall be registered as provided under section 11.8 below.
- 5. A person in whose name a Bond shall be registered shall (to the fullest extent permitted by law) be treated at all times and for all purposes as the absolute owner of such Bond regardless of any notice of ownership or trust.

11.3 Status and Security Interest

- The Bonds shall constitute the general, direct, unconditional and secured obligations of the Issuer. They
 shall at all times rank pari passu without any priority or preference among themselves. They shall rank
 subsequent to any other prior ranking indebtedness of the Company, if any.
- 2. In accordance with the provisions of the Trust Instrument, the Bonds shall be secured by the Security Interest held by the Security Trustee for the benefit of the Bondholders and in that respect only shall rank in preference to other present and future unsecured obligations of the Issuer, if any.
- 3. The ranking of the Initial Security Interest with respect to other lawful causes of preference is indicated in the Trust Instrument and in section 9.5 of this Securities Note.
- 4. In terms of article 1995 of the Civil Code (Cap. 16 of the laws of Malta), the property of a debtor is the common guarantee of his creditors, all of whom have an equal right over such property, unless there exist between them lawful causes of preference. Privileges and hypothecs are "lawful causes of preference". Accordingly any debts which are secured by privileges, whether existing now or which may come into existence in the future would rank according to the order set out by law, and debts which are secured by hypothecs would rank according to the date of registration, subject to what is provided in the Trust Instrument and in section 9.5 of the Securities Note.

11.4 Interest

- 1. The Bonds shall bear interest from and including 27 July 2016 at the rate of 4% per annum on the nominal value thereof, payable annually in arrears on each Interest Payment Date. The first interest payment will be effected on 27 July 2017 (covering the period 27 July 2016 to 26 July 2017).
- 2. Any Interest Payment Date which falls on a day other than a Business Day will be carried over to the next following day that is a Business Day. In terms of article 2156 of the Civil Code (Cap. 16 of the laws of Malta), the right of Bondholders to bring claims for payment of interest and repayment of the principal on the Bonds is barred by the lapse of five (5) years.
- 3. Each Bond will cease to bear interest from and including its due date of redemption unless payment of the principal in respect of the Bond is improperly withheld or refused or unless default is otherwise made in respect of payment, in which event interest shall continue to accrue at the applicable rate specified above or at the rate of two per cent (2%) per annum above the European Central Bank's Refinancing Rate, whichever is the greater.
- 4. When interest is required to be calculated for any period of less than a full year, it shall be calculated on the basis of a three hundred and sixty (360) day year consisting of twelve (12) months of thirty (30) days each, and in the case of an incomplete month, the number of days elapsed.

11.5 Payments

- 1. Payment of the principal amount (with interest accrued to the due date for redemption) as well as payment of any instalment of interest of the Bonds will be made in euro to the person in whose name such Bond is registered as at the close of business fifteen (15) days prior to the due date for redemption or fifteen (15) days prior to the Interest Payment Date (as the case may be) by direct credit to an account which is denominated in euro and held with any licensed bank in Malta as specified by the Bondholder. The Issuer shall not be responsible for any charges, loss or delay in transmission. Such payment shall be effected within seven (7) days of the due date for redemption or the Interest Payment Date (as the case may be).
- 2. All payments are subject in all cases to any pledge (duly constituted) of the Bonds and to any applicable fiscal or other laws and regulations. In particular, but without limitation, all payments by the Issuer in respect of the Bonds shall be made net of any amount which the Issuer is compelled by law to deduct or withhold for or on account of any present or future taxes, duties, assessments or other Governmental charges of whatsoever nature imposed or levied by or on behalf of the GOM or authority thereof or therein having power to tax.
- 3. No commissions or expenses shall be charged by the Issuer to the Bondholders in respect of payments made in accordance with this section 11.5. The Issuer shall not be liable for charges, expenses and commissions levied by parties other than the Issuer.

11.6 Redemption and Purchase

- 1. The Bonds will be redeemed at their nominal value (together with interest accrued to the due date for redemption) on 27 July 2026 (the "**Redemption Date**").
- 2. All Bonds so redeemed will be cancelled forthwith and may not be reissued or resold.
- 3. The Issuer and any Group Company may at any time purchase Bonds in the open market or otherwise at any price. Any purchase by tender shall be made available to all Bondholders alike. All Bonds so purchased by the Issuer will be cancelled forthwith and may not be re-issued or resold. All Bonds so purchased by any Group Company may be cancelled or held, reissued or resold at the discretion of the relevant purchaser. The Bonds so purchased, while held by or on behalf of any Group Company, shall not entitle the holder to vote at any meetings of the Bondholders and shall not be deemed to be outstanding for the purposes of these Bond Conditions including, without limitation, for the purposes of section 11.9.

11.7 Events of Default

- 1. The Security Trustee at its discretion may, and if so requested by a resolution passed by Bondholders holding not less than seventy-five per centum (75%) in nominal value of the Bonds then outstanding shall, give notice to the Issuer that the Bonds are, and they shall accordingly immediately become, due and repayable at their principal amount together with accrued interest, if any of the following events ("Events of Default") shall occur:-
 - (i) the Issuer shall fail to pay any interest on any Bond when due and such failure shall continue for sixty (60) days after written notice thereof shall have been given to the Issuer by the Trustee; or
 - (ii) the Issuer shall fail duly to perform or shall otherwise be in breach of any other material obligation contained in these Bond Conditions or in the Trust Instrument and such failure shall continue for sixty
 (60) days after written notice thereof shall have been given to the Issuer by the Security Trustee; or
 - (iii) any default occurs and continues for ninety (90) days under any contract or document relating to any Relevant Indebtedness (as defined below) of the Issuer in excess of €5,000,000 or its equivalent at any time; or
 - (iv) there shall have been entered against the Issuer a final judgment by a court of competent jurisdiction from which no appeal may be made or is taken for the payment of money in excess of €5,000,000 or its equivalent and ninety (90) days shall have passed since the date of entry of such judgment without its having been satisfied or stayed; or
 - (v) the Issuer is unable, or admits in writing its inability, to pay its debts as they fall due or otherwise becomes insolvent; or
 - (vi) an order is made or resolution passed or other action taken for the dissolution, termination of existence, liquidation, winding-up or bankruptcy of the Issuer; or
 - (vii) a provisional administrator is appointed of the whole or any part of the property of the Issuer or the Issuer ceases or threatens to cease to carry on its business or a substantial part thereof; or
 - (viii) all, or in the sole opinion of the Security Trustee, a material part, of the undertakings, assets, rights, or revenues of or shares or other ownership interests in the Issuer are seized, nationalised, expropriated or compulsorily acquired by or under the authority of any government;

provided that in the case of paragraphs (ii), (iii), (iv), (vii) and (viii) the Security Trustee shall have certified that in its opinion such event is materially prejudicial to the interests of the Bondholders.

- 2. At any time after the Bonds shall have become immediately due and payable as aforesaid, or in the event that the Issuer fails to pay the principal amount as and when the Bonds are due to be redeemed as provided by these Bond Conditions, and such failure shall continue for thirty (30) days after written notice thereof shall have been given to the Company, the Security Trustee may, at its discretion, and shall, if so requested by resolution passed by Bondholders holding not less than seventy-five per centum (75%) in nominal value of the Bonds then still outstanding, institute such proceedings as it may think fit against the Issuer to enforce repayment of the principal together with accrued but unpaid interest, including the enforcement of the Security Interest, provided that the Trustee shall not be bound to do so unless it shall have been indemnified to its satisfaction against all actions, proceedings, claims and demands to which it may thereby render itself liable and all costs, charges, damages and expenses which it may incur by so doing.
- 3. The Security Trustee shall not be bound to take any steps to ascertain whether any Event of Default or other condition, event or circumstance has occurred or may occur, and, until it shall have actual knowledge or express notice to the contrary, the Security Trustee shall be entitled to assume that no such Event of Default or condition, event or other circumstance has happened and that the Issuer is observing and performing all the obligations, conditions and provisions on their respective parts contained in the Bond Conditions and the Trust Instrument.
- 4. No Bondholder shall in any circumstances be entitled to any remedy (whether by way of action or otherwise howsoever) for the recovery of the Bonds or any part of it or any interest, unless the Security Trustee, having

become bound to take proceedings in accordance with the Trust Instrument, fails to do so. In that case any Bondholder may, on giving the Security Trustee an indemnity satisfactory to it, in the name of the Security Trustee (but not otherwise) himself either take such proceedings against the Issuer or prove in the windingup of the Issuer.

- For the purposes of this section 11.7, "Relevant Indebtedness" means any indebtedness in respect of:
 - (i) monies borrowed:
 - (ii) any debenture, bond, note, loan stock or other security creating or acknowledging indebtedness;
 - (iii) any acceptance credit;
 - (iv) the acquisition cost of any asset to the extent payable before or after the time of acquisition or possession by the party liable where the advance or deferred payment is arranged primarily as a method of raising finance or financing the acquisition of that asset;
 - (v) leases entered into primarily as a method of raising finance or financing the acquisition of the asset
 - (vi) amounts raised under any other transaction having the commercial effect of borrowing or raising of money;
 - (vii) any guarantee, indemnity or similar assurance in respect of any such indebtedness.

11.8 Registration, Replacement, Transfer and Exchange

- A register of the Bonds will be kept wherein there will be entered the names and addresses of the Bondholders and particulars of the Bonds held by them respectively and a copy of such register will at all reasonable times during business hours be open to the inspection of the Bondholders at the registered office of the Issuer for the purpose of the Bondholders inspecting information held on their respective account. If and for as long as the Bonds are admitted to listing on the MSE, the said register will be kept at the Central Securities Depository of the MSE or as may be stipulated by, and in accordance with, the MSE Bye-Laws. The said CSD will issue on a twelve-monthly basis or at such other intervals as the MSE Bye-Laws shall from time to time determine, a statement of holdings to Bondholders evidencing their entitlement to Bonds held in the register kept by the said CSD.
- 2. A Bond may be transferred or transmitted only in whole, in multiples of €100, in accordance with the rules and procedures applicable from time to time at the MSE.
- Any person becoming entitled to a Bond in consequence of the death, bankruptcy or winding-up of a Bondholder may, upon such evidence being produced as may from time to time properly be required by the Issuer or the MSE, elect either to be registered himself as Bondholder or to have some person nominated by him registered as the transferee thereof. If the person so becoming entitled shall elect to be registered himself, he shall deliver or send to the CSD a notice in writing signed by him stating that he so elects. If he shall elect to have another person registered he shall testify his election by transferring the Bond, or procuring the transfer of the Bond, in favour of that person.
- 4. All transfers and transmissions are subject in all cases to any pledge (duly constituted) of the Bonds and to any applicable laws and regulations.
- The cost and expenses of effecting any exchange or registration of transfer or transmission except for the expenses of delivery by regular mail (if any) and except, if the Issuer shall so require, the payment of a sum sufficient to cover any tax, duty or other governmental charge or insurance charges that may be imposed in relation thereto, will be borne by the Bondholder.
- The Issuer will not register the transfer or transmission of Bonds for a period of fifteen (15) days preceding the due date for any payment of interest on the Bonds or the due date for redemption.

11.9 Resolutions and Meetings of Bondholders

- 1. The Issuer may, through the Security Trustee, from time to time call meetings of Bondholders for the purpose of consultation with Bondholders or for the purpose of obtaining the consent of Bondholders on matters which, in terms of the Prospectus, require the approval of a meeting of Bondholders and to effect any change to the applicable Bond Conditions, including any change to a material term of issuance of the Bonds or the Prospectus, or to effect a change to the Trust Instrument.
 - (i) No change to the applicable Bond Conditions or any terms of this Prospectus which has an effect on the Bonds may be made unless such decision is made at a meeting of Bondholders duly convened and held.
 - (ii) A meeting of Bondholders shall be called by the Directors, through the Security Trustee, by giving all Bondholders listed on the register of Bondholders as at a date being not more than thirty (30) days preceding the date scheduled for the meeting, not less than fourteen (14) days' notice in writing. Such notice shall set out the time, place and date set for the meeting and the matters to be discussed or decided thereat, including, if applicable, sufficient information on any amendment of the Prospectus that is proposed to be voted upon at the meeting and seeking the approval of the Bondholders. Following a meeting of Bondholders held in accordance with the provisions contained hereunder, the Issuer shall, acting in accordance with the resolution(s) taken at the meeting, communicate to the Bondholders whether the necessary consent to the proposal made by the Issuer has been granted or withheld. Subject to having obtained the necessary approval by the Bondholders in accordance with the provisions of this section 11.9 at a meeting called for that purpose as aforesaid, any such decision shall subsequently be given effect to by the Issuer.
 - (iii) The amendment or waiver of any of the provisions of and/or conditions contained in this Securities Note, or in any other part of the Prospectus, may only be made with the approval of Bondholders at a meeting called and held for that purpose in accordance with the terms hereof.
 - (iv) A meeting of Bondholders shall only validly and properly proceed to business if there is a quorum present at the commencement of the meeting. For this purpose at least two (2) Bondholders present, in person or by proxy, representing not less than 50% in nominal value of the Bonds then outstanding, shall constitute a quorum. If a quorum is not present within thirty (30) minutes from the time scheduled for the commencement of the meeting as indicated on the notice convening same, the meeting shall stand adjourned to a place, date and time as shall be communicated by the Directors to the Bondholders present at that meeting. The Issuer shall within two (2) days from the date of the original meeting publish by way of a company announcement the date, time and place where the adjourned meeting is to be held. An adjourned meeting shall be held not earlier than seven (7) days, and not later than fifteen (15) days, following the original meeting. At an adjourned meeting: the number of Bondholders present, in person or by proxy, shall constitute a quorum; and only the matters specified in the notice calling the original meeting shall be placed on the agenda of, and shall be discussed at, the adjourned meeting.
 - (v) Any person who in accordance with the Memorandum and Articles of Association is to chair the annual general meetings of shareholders shall also chair meetings of Bondholders.
 - (vi) Once a quorum is declared present by the chairman of the meeting, the meeting may then proceed to business and address the matters set out in the notice convening the meeting. In the event of decisions being required at the meeting the Directors or their representative shall present to the Bondholders the reasons why it is deemed necessary or desirable and appropriate that a particular decision is taken. The meeting shall allow reasonable and adequate time to Bondholders to present their views to the Issuer and the other Bondholders present at the meeting. The meeting shall then put the matter as proposed by the Issuer to a vote of the Bondholders present at the time at which the vote is being taken, and any Bondholders taken into account for the purpose of constituting a quorum who are no longer present for the taking of the vote shall not be taken into account for the purpose of such vote.
 - (vii) The voting process shall be managed by the Company Secretary under the supervision and scrutiny of the auditors of the Issuer and the Security Trustee.
 - (viii) The proposal placed before a meeting of Bondholders shall only be considered approved if at least 60% in nominal value of the Bondholders present at the meeting at the time when the vote is being taken, in person or by proxy, shall have voted in favour of the proposal.

- (ix) Save for the above, the rules generally applicable to proceedings at general meetings of shareholders of the Issuer shall *mutatis mutandis* apply to meetings of Bondholders.
- Without prejudice to the above, the Security Trustee must convene a meeting of the Bondholders upon a requisition in writing signed by a Bondholder or Bondholders holding not less than twenty per cent (20%) in nominal amount of the beneficial interest in the Bonds for the time being outstanding and upon receiving such indemnity against the costs of convening and holding such meeting as it may reasonably require. Meetings shall be convened and requested only for the purpose of considering a resolution specified in the immediately preceding paragraph. The time and place of the meeting must be specified by the Security Trustee. Subject to what is stated in this paragraph, the rules applicable to proceedings at general meetings of Bondholders specified in the immediately preceding paragraph shall mutatis mutandis apply to meetings of Bondholders so convened.
- 3. The Security Trustee at any time, prior to exercising any power or discretion under the Trust Instrument may call a meeting of Bondholders.
- 4. In the event that the Bonds are listed, and for so long as the Bonds remain listed, any resolution which may be proposed shall be subject to any laws, regulations, rules or bye-laws which may be applicable from time to time.

11.10 Further Issues

The Issuer may, from time to time, without the consent of the Bondholders, create and issue further bonds, debentures or any other debt securities either having the same terms and conditions as the Bonds in all respects (except for the first payment of interest on them) and so that such further issue shall be consolidated and form a single series with the Bonds or otherwise upon such terms and conditions as the Issuer may determine. Any further debt securities so issued may rank *pari passu* in all respects with the Bonds but shall not rank ahead of the Bonds.

11.11 Bonds held Jointly

In respect of a Bond held jointly by several persons (including husband and wife), the joint holders shall nominate one of their number as their representative and his/her name will be entered in the register with such designation. The person whose name shall be inserted in the field entitled "Applicant" on the Application Form, or first named in the register of Bondholders shall for all intents and purposes be deemed to be such nominated person by all those joint holders whose names appear in the field entitled "Additional Applicants" in the Application Form or joint holders in the register as the case may be. Such person shall, for all intents and purposes, be deemed to be the registered holder of the Bond/s so held.

11.12 Bonds held Subject to Usufruct

In respect of a Bond held subject to usufruct, the name of the bare owner and the usufructuary shall be entered in the register. The usufructuary shall, for all intents and purposes, be deemed *vis-a-vis* the Issuer to be the holder of the Bond/s so held and shall have the right to receive interest on the Bond/s and to vote at meetings of the Bondholders but shall not, during the continuance of the Bond/s, have the right to dispose of the Bond/s so held without the consent of the bare owner.

11.13 Governing Law and Jurisdiction

- 1. The Bonds are governed by and shall be construed in accordance with Maltese law.
- 2. Any legal action, suit or proceeding against the Issuer arising out of or in connection with the Bonds shall be brought exclusively before the Maltese Courts and the Bondholders shall, upon and by submitting an Application, acknowledge that they are submitting to the exclusive jurisdiction of the Maltese Courts as aforesaid.

11.14 Notices

Notices will be mailed to Bondholders at their registered addresses and shall be deemed to have been served at the expiration of twenty-four (24) hours after the letter containing the notice is posted, and in proving such service it shall be sufficient to prove that a prepaid letter containing such notice was properly addressed to such Bondholders at his/her/its registered address and posted.

11.15 Listing

The Bonds, upon issue and subscription, shall be admitted to the Official List of the MSE. Accordingly, all these Bond Conditions shall be read in conjunction with the Listing Rules of the Listing Authority and with the Bye-Laws of the MSE as applicable from time to time.

12. TERMS AND CONDITIONS OF THE APPLICATION

The following are the terms and conditions which are applicable to Applications (the "**Application Conditions**"). Any Applicant is deemed to have notice of, and is bound by, these Application Conditions.

- 1. The issue and allotment of the Bonds is conditional upon: (i) the Bonds being admitted to the Official List of the MSE; and (ii) the Initial Security Interest being constituted in favour of the Security Trustee. In the event that either of the aforesaid conditions is not satisfied within 30 Business Days of the close of the Public Offer Period, any Application monies received by the Issuer will be returned without interest by direct credit into the Applicant's bank account indicated by the Applicant on the relative Application Form.
- 2. Subject to all other Application Conditions, the Issuer reserves the right to reject, in whole or in part, or to scale down any Application, including multiple or suspected multiple Applications, and to present any cheques and/or drafts for payment upon receipt. The right is also reserved to refuse any Application, which in the opinion of the Issuer is not properly completed in all respects in accordance with the instructions or is not accompanied by the required documents.
- 3. If any Application is not accepted, or if any Application is accepted for fewer Bonds than those applied for, then the Applicant shall receive a refund of the price of the Bonds applied for but not allocated, the Application monies or the balance of the amount paid on Application will be returned by direct credit into the Applicant's bank account as indicated in the Application Form, within five (5) Business Days from the date of final allocation. No interest shall be due on refunds and the Issuer shall not be responsible for any charges, loss or delay in transmission.
- 4. In the case of joint Applications, reference to the Applicant in these Application Conditions is a reference to each Applicant, and liability therefor is joint and several.
- 5. The consideration payable by holders of EUR Bonds 2016/18 applying for Bonds may be settled, after submitting the appropriate Application Form 'A', by the transfer to the Issuer of all or part of the EUR Bonds 2016/18 held by such Applicant as at the Cut-Off Date, subject to a minimum application of €2,000, which transfer shall be effected at the par value of the EUR Bonds 2016/18. Any holders of EUR Bonds 2016/18 whose holding in EUR Bonds 2016/18 is less than €2,000 shall be required to pay the difference together with the submission of their Application Form 'A'.

Holders of EUR Bonds 2016/18, electing to subscribe for Bonds through the Bond Exchange Programme, shall be allocated Bonds for the corresponding nominal value of EUR Bonds 2016/18 transferred to the Issuer (including cash top-up, where applicable).

The consideration payable by holders of GBP Bonds 2016/18 applying for Bonds may be settled, after submitting the appropriate Application Form 'B', by the transfer to the Issuer of all or part of the GBP Bonds 2016/18 held by such Applicant as at the Cut-Off Date. For such purpose, the par value of each holding in GBP Bonds 2016/18 shall be converted to the equivalent value in euro at the exchange rate established by the Issuer on 27 June 2016 as described in section 8.5 above. Subscriptions shall be subject to a minimum application of €2,000 in Bonds. Any holders of GBP Bonds 2016/18 whose holding in GBP Bonds 2016/18 is less than the equivalent amount of €2,000 shall be required to pay the difference in euro together with the submission of their Application Form 'B'.

Holders of GBP Bonds 2016/18, electing to subscribe for Bonds through the Bond Exchange Programme, shall be allocated Bonds for the corresponding equivalent euro amount of nominal value of GBP Bonds 2016/18 transferred to the Issuer (including cash top-up payable in euro, where applicable).

The transfer of Bonds 2016/18 to the Issuer in consideration for the subscription for Bonds shall cause the obligations of the Issuer with respect to such Bonds 2016/18 to be extinguished, and shall give rise to obligations on the part of the Issuer under the Bonds.

Bonds applied for by Existing Holders through the Bond Exchange Programme shall be allocated prior to any other allocation of Bonds.

In addition to the aforesaid, Existing Holders transferring all of the Bonds 2016/18 held by them as at the Cut-Off Date, may apply for an amount of Bonds in excess of the amount of Bonds 2016/18 being transferred. In such a case Existing Holders may subscribe for additional Bonds in multiples of €100 by completing the appropriate section of the respective Application Form 'A' and/or Application Form 'B'.

By virtue of the submission of the duly completed and signed Application Form 'A' and/or Application Form 'B', Existing Holders shall be deemed:

- (i) to confirm that all or part (as the case may be) of the said Bonds 2016/18 as indicated in the Application Form are to be transferred in the Issuer's favour for cancellation; and
- (ii) as an irrevocable mandate to the Issuer to engage the services of such brokers or intermediaries as may be necessary to fully and effectively carry out all procedures necessary and to fully and effectively vest title in the appropriate number of Bonds in the Existing Holder.
- 6. Any person, whether natural or legal, shall be eligible to submit an Application, and any one person, whether directly or indirectly, should not submit more than one Application Form, provided that an Application Form submitted in terms of Condition 4 above shall not be counted for this purpose. In the case of corporate Applicants or Applicants having separate legal personality, the Application Form must be signed by a person authorised to sign on behalf of, and bind, such Applicant. It shall not be incumbent on the Issuer or Registrar to verify whether the person or persons purporting to bind such an Applicant is or are in fact authorised.
- 7. Applications in the name and for the benefit of minors shall be allowed provided that they are signed by both parents or by the legal guardian/s and accompanied by a Public Registry birth certificate of the minor in whose name and for whose benefit the Application Form is submitted. Any Bonds allocated pursuant to such an Application shall be registered in the name of the minor as Bondholder, with interest payable to the parents or legal guardian/s signing the Application Form until such time as the minor attains the age of eighteen (18) years, following which all interests shall be payable directly to the registered holder, provided that the Issuer has been duly notified in writing of the fact that the minor has attained the age of eighteen (18) years.
- 8. Holders of EUR Bonds 2016/18, holders of GBP Bonds 2016/18, Current Shareholders and the general public may subscribe for Bonds by submitting Application Form 'A', Application Form 'B', Application Form 'C' and Application Form 'D' respectively.
- 9. Application Form 'A', Application Form 'B' and Application Form 'C' are to be completed and submitted by not later than 10:00 hours on 18 July 2016. Subscription lists, in relation to the general public offer, will open at 08:30 hours on 18 July 2016 and will close thereafter as may be determined by the Issuer, but in any event not later than 16:00 hours on 20 July 2016.
- 10. The minimum subscription of the Bonds is €2,000 in value. Applications in excess of the said minimum thresholds must be in multiples of €100. The completed Application Forms are to be lodged with any of the Authorised Financial Intermediaries mentioned in this Prospectus. Submission of Application Form 'A' and Application Form 'B' by Existing Holders must be accompanied by payment representing difference between the full price of the amount of Bonds applied for and the nominal value, or the euro equivalent amount as the case may be, of Bonds 2016/18 being surrendered. Submission of Application Form 'C' by Current Shareholders and Application Form 'D' by the general public must be accompanied by the full price of the Bonds applied for in euro. Payment may be made either in cash or by cheque payable to "The Registrar MIDI Bond Issue 2016". In the event that a cheque accompanying an Application Form is not honoured on its first presentation, the Issuer and the Registrar reserve the right to invalidate the relative Application Form.

- By completing and delivering an Application Form you (as the Applicant(s)):
 - (i) irrevocably offer to purchase the number of Bonds specified in your Application Form (or any smaller number for which the Application is accepted) at the Issue Price subject to the Prospectus, the Bond Conditions and the Memorandum and Articles of Association;
 - (ii) authorise the Registrar and the Directors of the Issuer to include your name or in the case of joint Applications, the first named Applicant, in the register of Debentures of the Issuer in respect of the Bonds allocated to you;
 - (iii) warrant that your remittance will be honoured on first presentation and agree that, if such remittance is not so honoured, you will not be entitled to receive a registration advice, or to be registered in the register of Debentures or to enjoy or receive any rights in respect of such Bonds unless and until you make payment in cleared funds for such Bonds and such payment is accepted by the Issuer (which acceptance shall be made in its absolute discretion and may be on the basis that you indemnify it against all costs, damages, losses, expenses and liabilities arising out of or in connection with the failure of your remittance to be honoured on first presentation) and that, at any time prior to unconditional acceptance by the Issuer of such late delivery of consideration in respect of such Bonds, the Issuer may (without prejudice to other rights) treat the agreement to allocate such Bonds as void and may allocate such Bonds to some other person, in which case you will not be entitled to any refund or payment in respect of such Bonds (other than return of such late payment);
 - (iv) agree that the registration advice and other documents and any monies returnable to you may be retained pending clearance of your remittance and any verification of identity as required by the Prevention of Money Laundering Act (Cap. 373 of the laws of Malta), and regulations made thereunder, and that such monies will not bear interest;
 - (v) agree that all Applications, acceptances of Applications and contracts resulting therefrom will be governed by, and construed in accordance with Maltese law and that you submit to the jurisdiction of the Maltese Courts and agree that nothing shall limit the right of the Issuer to bring any action, suit or proceeding arising out of or in connection with any such Applications, acceptances of Applications and contracts in any other manner permitted by law in any court of competent jurisdiction;
 - (vi) warrant that, if you sign the Application Form on behalf of another party or on behalf of a corporation or corporate entity or association of persons, you have due authority to do so and such person, corporation, corporate entity, or association of persons will also be bound accordingly and will be deemed also to have given the confirmations, warranties and undertakings contained in these Application Conditions and undertake to submit your power of attorney or a copy thereof duly certified by a lawyer or notary public if so required by the Registrar;
 - (vii) agree that all documents in connection with the issue of the Bonds and any returned monies including refund of all unapplied Application monies will be returned at your risk and will be returned by direct credit into the bank account as specified in the Application Form;
 - (viii) agree that, having had the opportunity to read the Prospectus, you have, and shall be deemed to have had, notice of all information and representations concerning the Issuer and the issue of the Bonds contained therein:
 - (ix) confirm that in making such Application you are not relying on any information or representation in relation to the Issuer or the issue of the Bonds other than those contained in the Prospectus and you accordingly agree that no person responsible solely or jointly for the Prospectus or any part thereof will have any liability for any such other information or representation;
 - (x) confirm that you have reviewed and you will comply with the restriction contained in Condition 12 below;
 - (xi) warrant that you are not under the age of eighteen (18) years or if you are lodging an Application in the name and for the benefit of a minor, warrant that you are the parents or legal guardian/s of the minor;
 - (xii) agree that such Application Form is addressed to the Issuer and that in respect of those Bonds for which your Application has been accepted, you shall receive a registration advice confirming such acceptance;

- (xiii) confirm that in the case of a joint Application the first-named Applicant shall be deemed the holder of the Bonds;
- (xiv) agree to provide the Registrar and/or Issuer as the case may be, with any information which it may request in connection with your Application(s);
- (xv) agree that Charts Investment Management Service Limited will not, in its capacity of Sponsor, treat you as its customer by virtue of your making an Application for the Bonds and that Charts Investment Management Service Limited will owe you no duties or responsibilities concerning the price of the Bonds or their suitability for you;
- (xvi) warrant that, in connection with your Application, you have observed all applicable laws, obtained any requisite governmental or other consents, complied with all requisite formalities and paid any issue, transfer or other taxes due in connection with your Application in any territory and that you have not taken any action which will or may result in the Issuer or the Registrar acting in breach of the regulatory or legal requirements of any territory in connection with the Bond Issue or your Application;
- (xvii) warrant that all applicable exchange control permits and authorisations which may be applicable have been duly and fully complied with;
- (xviii) represent that you are not a U.S. person (as such term is defined in Regulation "S" under the U.S. Securities Act of 1933, as amended (the "Securities Act, 1933")) and that you are not accepting the invitation comprised in the Prospectus from within the United States of America, its territories or its possessions, any State of the United States of America or any area subject to its jurisdiction (the "United States") or on behalf or for the account of anyone within the United States or anyone who is a U.S. person, unless you indicate otherwise on the Application Form in accordance with the instructions of the Application Form.
- 12. The Bonds have not been and will not be registered under the Securities Act, 1933 and accordingly may not be offered or sold within the United States or to or for the account or benefit of a U.S. person.
- 13. No person receiving a copy of the Prospectus or an Application Form in any territory other than Malta may treat the same as constituting an invitation or offer to him nor should he in any event use such Application Form, unless, in the relevant territory, such an invitation or offer could lawfully be made to him or such Application Form could lawfully be used without contravention of any registration or other legal requirements. It is the responsibility of any person outside Malta wishing to make any Application to satisfy himself as to full observance of the laws of any relevant territory in connection therewith, including obtaining any requisite governmental or other consents, observing any other formalities required to be observed in such territory and paying any issues, transfer or other taxes required to be paid in such territory.
- 14. For the purposes of the Prevention of Money Laundering and Funding of Terrorism Regulations 2008 (Legal Notice 180 of 2008, as subsequently amended), all Authorised Financial Intermediaries are under a duty to communicate, upon request, all information they hold about clients, pursuant to Articles 1.2(d) and 2.4 of the "Members' Code of Conduct" appended as Appendix 3.6 of the MSE Bye-Laws, irrespective of whether the Authorised Financial Intermediaries are Exchange members or not. Such information shall be held and controlled by the Malta Stock Exchange in terms of the Data Protection Act (Cap. 440 of the laws of Malta) for the purposes, and within the terms of, the Malta Stock Exchange's data protection and privacy policy as published from time to time.
- 15. Within five (5) Business Days from the closing of the subscription lists, the Issuer shall determine and announce by way of a company announcement, the basis of acceptance of Applications and allocation policy to be adopted.
- 16. Save where the context requires otherwise, terms defined in the Prospectus bear the same meaning when used in these Application Conditions, in the Application Form and in any other document issued pursuant to the Prospectus.

Annex I Specimen Application Forms



MIDI P.L.C. €50,000,000 4% SECURED BONDS 2026 APPLICATION FORM 'A' **EUR BONDS 2016/18**

Please read the notes overleaf before completing this Application Form. Marl	k 'X' if applicable.	
APPLICANT (See notes 2 & 4)		
	TEL. NO.	MOBILE NO.
This Application Form is not transferable and entitles you to a preferential trea and is to be submitted as a method of payment where the Applicant selects as to transfer to the Issuer all or part of the holding in EUR Bonds 2016/18 is set out in Box 1 of Panel B hereunder. By submitting this signed Applicat i. to confirm that all or part (as the case may be) of the said Bonds 201 Issuer's favour for cancellation; and ii. as an irrevocable mandate to the Issuer to engage the services of such carry out all procedures necessary and to fully and effectively vest title in	s to apply for the Midi p.l.c. 49 neld by the Applicant as at the ion Form, Existing Holders shi 6/18 as indicated in the Applen brokers or intermediaries as	% Secured Bonds 2026 (the "Bonds"): Cut-Off Date, the nominal value of white all be deemed: lication Form are to be transferred in the may be necessary to fully and effective
I/WE APPLY TO PURCHASE AND/OR ACQUIRE (See not	te 6)	
BOX 1 - Nominal Value of the EUR Bonds 2016/18.		AMOUNT IN FIGURES Box 1
BOX 2 - I/We wish to purchase and acquire the amount set out in Box 2 Price (at par) pursuant to the Prospectus dated 28 June 2016 (minimum €100 thereafter).		€
AMOUNT IN WORDS		AMOUNT IN FIGURES Box 2
		€
		AMOUNT IN FIGURES Box 3
BOX 3 - Amount of Bonds applied for less the nominal holding in EUR Bo	ande 2016/18 navable in full	Difference payable on Application
	mus 2010/10 payable in full	Box 2 - Box 1
upon application under the Terms and Conditions of the Bonds set out in the	ne Prospectus.	Box 2 - Box 1 €
upon application under the Terms and Conditions of the Bonds set out in the	ne Prospectus.	Box 2 - Box 1 €
upon application under the Terms and Conditions of the Bonds set out in the RESIDENT - WITHHOLDING TAX DECLARATION (See no	ne Prospectus.	
upon application under the Terms and Conditions of the Bonds set out in the RESIDENT - WITHHOLDING TAX DECLARATION (See no.	ne Prospectus. otes 7 & 8a) (to be comple	€
upon application under the Terms and Conditions of the Bonds set out in the RESIDENT - WITHHOLDING TAX DECLARATION (See not live elect to have final withholding tax deducted from my/our interest.	otes 7 & 8a) (to be comple	€
upon application under the Terms and Conditions of the Bonds set out in the RESIDENT - WITHHOLDING TAX DECLARATION (See no.	otes 7 & 8a) (to be comple	€
upon application under the Terms and Conditions of the Bonds set out in the RESIDENT - WITHHOLDING TAX DECLARATION (See not live elect to have final withholding tax deducted from my/our interest.	ntes 7 & 8a) (to be complete.) Iding tax).	€ eted ONLY if the Applicant is a Resident of Ma
RESIDENT - WITHHOLDING TAX DECLARATION (See no I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withholding tax purposes) NON-RESIDENT - DECLARATION FOR TAX PURPOSES (See no I/We are the content of the	ntes 7 & 8a) (to be complete.) Iding tax).	€ eted ONLY if the Applicant is a Resident of Ma
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the line) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY) T.I.N. (Tax Identification Number)	ne Prospectus. Ito be comple Iding tax). See notes 3,7,8 & 8a) (to be cor CITY OF BIRTH COUNTRY OF BIRTH	€ eted ONLY if the Applicant is a Resident of Ma
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the line) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY) T.I.N. (Tax Identification Number)	otes 7 & 8a) (to be compled by the second by	€ eted ONLY if the Applicant is a Resident of Ma
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the line) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY) T.I.N. (Tax Identification Number)	ne Prospectus. Ito be completed from the completed from the conformation of the confo	€ eted ONLY if the Applicant is a Resident of Ma
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withholding tax COUNTRY TAX COUNTRY T.I.N. (Tax Identification Number) PASSPORT/NATIONAL I.D. CARD NUMBER COUNTRY COU	ne Prospectus. Ito be completed to be completed to be completed to be completed to be concerned to be concern	€ eted ONLY if the Applicant is a Resident of Me mpleted ONLY if the Applicant is a Non-Reside
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the line) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold non-resident - Declaration for Tax purposes (STAX COUNTRY (Tax Identification Number) (COUNTRY (COUNT	ne Prospectus. Ito be completed to be completed to be completed to be completed to be concerned to be concern	€ eted ONLY if the Applicant is a Resident of Me mpleted ONLY if the Applicant is a Non-Reside
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the line) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold non-resident - Declaration for Tax purposes (STAX COUNTRY (Tax Identification Number) (COUNTRY (COUNT	ne Prospectus. Ito be completed in the second in the seco	eted ONLY if the Applicant is a Resident of Management of Management (1997) et al. (1
RESIDENT - WITHHOLDING TAX DECLARATION (See not in the later of the la	ne Prospectus. Ito be completed in the second in the seco	eted ONLY if the Applicant is a Resident of Management of Management of Management of the Applicant is a Non-Resident of Management of Manage
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (COUNTRY (otes 7 & 8a) (to be completed of the state o	eted ONLY if the Applicant is a Resident of Management of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of the Applicant is a Resident of Management of the Applicant is a Resident of Management of the Applicant is a Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of the Applica
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (T.I.N. (Tax Identification Number) (PASSPORT/NATIONAL I.D. CARD NUMBER (COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See no BANK) I/We have fully understood the instructions for completing this Application in the Interest of the property of the pr	ne Prospectus. Ito be completed a service of the service of the completed and the content of th	eted ONLY if the Applicant is a Resident of Management of Management of Management of the Applicant is a Non-Resident of the Applicant is a Non-Resident of Management of the Applicant is a Resident of Management
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (COUNTRY (ne Prospectus. Ito be completed a service of the service of the completed and the content of th	eted ONLY if the Applicant is a Resident of Me mpleted ONLY if the Applicant is a Non-Reside ISSUE DATE Completion of this Panel is MANDATOR g this Application solely on the basis
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (T.I.N. (Tax Identification Number) (PASSPORT/NATIONAL I.D. CARD NUMBER (COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See no BANK) I/We have fully understood the instructions for completing this Application in the Interest of the property of the pr	ne Prospectus. Ito be completed a service of the service of the completed and the content of th	eted ONLY if the Applicant is a Resident of Me mpleted ONLY if the Applicant is a Non-Reside ISSUE DATE Completion of this Panel is MANDATOR g this Application solely on the basis
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (T.I.N. (Tax Identification Number) (PASSPORT/NATIONAL I.D. CARD NUMBER (COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See no BANK) I/We have fully understood the instructions for completing this Application in the Interest of the property of the pr	ne Prospectus. Ito be completed a service of the service of the completed and the content of th	eted ONLY if the Applicant is a Resident of Management of Management of Management of the Applicant is a Non-Resident of the Applicant is a Non-Resident of Management of the Applicant is a Resident of Management
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY (CTAX Identification Number) (CTAX Identification Number) PASSPORT/NATIONAL I.D. CARD NUMBER (COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See no BANK IBAN I/We have fully understood the instructions for completing this Application the Prospectus, and subject to its Terms and Conditions of the Bonds (a)	iding tax). See notes 3,7,8 & 8a) (to be completed in the completed in th	eted ONLY if the Applicant is a Resident of Management of Management of Management of ONLY if the Applicant is a Non-Resident of Management of the Applicant is a Resident of Management of Managemen
RESIDENT - WITHHOLDING TAX DECLARATION (See note of location) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold non-resident - Declaration for tax purposes (STAX COUNTRY (T.I.N. (Tax Identification Number) (PASSPORT/NATIONAL I.D. CARD NUMBER (COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See note) BANK (IBAN) I/We have fully understood the instructions for completing this Application the Prospectus, and subject to its Terms and Conditions of the Bonds (a Signature/s of Applicant/s (Parent/s or legal guardian/s are/is to sign if Applicant is a minor)	iding tax). See notes 3,7,8 & 8a) (to be completed in the completed in th	eted ONLY if the Applicant is a Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of the Applicant is a Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of Management of the Applicant is a Non-Resident of the Applicant is a Non-Res
RESIDENT - WITHHOLDING TAX DECLARATION (See not) I/We elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY C.I.N. (Tax Identification Number) PASSPORT/NATIONAL I.D. CARD NUMBER COUNTRY I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See no BANK IBAN I/We have fully understood the instructions for completing this Application the Prospectus, and subject to its Terms and Conditions of the Bonds (a Signature/s of Applicant/s	inters 7 & 8a) (to be completed of 8 & 8a) (to be completed of 8 & 8a) (to be concerned of 8 & 8a) (to be completed of 8 & 8a) (to be concerned of 8 & 8a) (to be concerne	eted ONLY if the Applicant is a Resident of Manual
RESIDENT - WITHHOLDING TAX DECLARATION (See note of live elect to have final withholding tax deducted from my/our interest. I/We elect to receive interest GROSS (i.e. without deduction of withhold non-resident of my/our interest.) NON-RESIDENT - DECLARATION FOR TAX PURPOSES (STAX COUNTRY T.I.N. (Tax Identification Number) PASSPORT/NATIONAL I.D. CARD NUMBER COUNTR I/We am/are NOT Resident in Malta but I/we am/are Resident in the EI/We am/are NOT Resident in Malta and I/we am/are NOT Resident in INTEREST, REFUND & REDEMPTION MANDATE (See note) BANK IBAN I/We have fully understood the instructions for completing this Application the Prospectus, and subject to its Terms and Conditions of the Bonds (a) Signature/s of Applicant/s are/is to sign if Applicant is a minor) (All parties are to sign in the case of a joint Application) (Bare owner/s and usufructuary/ies to sign in the case of holdings of EUR Bonds 2016)	inters 7 & 8a) (to be completed of 8 & 8a) (to be completed of 8 & 8a) (to be concerned of 8 & 8a) (to be completed of 8 & 8a) (to be concerned of 8 & 8a) (to be concerne	eted ONLY if the Applicant is a Resident of Manual

Notes on how to complete this Application Form and other information

The following notes are to be read in conjunction with the Prospectus dated 28 June 2016 regulating the Bond Issue

- This Application is governed by the Terms and Conditions of the Application contained in Section 12 of the Securities Note dated 28 June 2016
 forming part of the Prospectus. Capitalised terms not defined herein shall, unless the context otherwise requires, have the meaning ascribed to
 them in the Prospectus.
- 2. This Application Form is to be completed in BLOCK LETTERS.
- 3. Applicants who are Non-Residents in Malta for tax purposes must complete Panel D overleaf.
- 4. The MSE account number pertaining to the Applicant has been pre-printed in Panel A and reflects the MSE account number on the Issuer's Register at the CSD as at 24 June 2016 (trading session of the 22 June 2016). APPLICANTS ARE TO NOTE THAT ANY BONDS ALLOTTED TO THEM WILL BE RECORDED BY THE MALTA STOCK EXCHANGE IN THE MSE ACCOUNT QUOTED ON THIS APPLICATION FORM EVEN IF THE DETAILS OF SUCH MSE ACCOUNT, AS HELD BY THE CSD OF THE MALTA STOCK EXCHANGE, DIFFER FROM ANY OR ALL OF THE DETAILS APPEARING OVERLEAF.
- 5. In the case where a holder of Bonds 2016/18 is a body corporate, Application Forms must be signed by duly authorised representative/s indicating the capacity in which they are signing.
- 6. The amount set out in Box 2 of Panel B overleaf must be in multiples of €100. The Issuer will be giving preference to Applications made by holders of EUR Bonds 2016/18 up to their full amount held as at the Cut-Off Date, subject to a minimum application of €2,000.
 - Where the Applicant wishes to acquire a number of Bonds having an aggregate value which exceeds the nominal value of the number of EUR Bonds 2016/18 set out in Box 1 of Panel B, the Applicant may do so by including such higher amount in Box 2 in Panel B. In such case, the Applicant must ensure that the relative Application Form is accompanied by payment of the difference between the full price of the amount of Bonds applied for and the nominal value of Bonds 2016/18 being transferred. Payment of the amount representing such difference, which is to be inserted in Box 3 of Panel B overleaf, must be made in Euro in cleared funds to "The Registrar Midi Bond Issue 2016". In the event that the cheque accompanying an Application Form is not honoured on the first presentation, the Issuer and the Registrar reserve the right to invalidate the relative Application.
- 7. Only Applicants who hold a valid official Maltese Identity Card or companies registered in Malta will be treated as residents in Malta. In such a case, the Applicant may elect to have final withholding tax, currently 15%, deducted from interest payments in which case, such interest need not be declared in the Applicant's income tax return. The Applicant may elect to receive the interest gross (i.e. without deduction of final withholding tax), but will then be obliged to declare interest so received in the tax return. Interest received by non-resident Applicants is not taxable in Malta and non-residents will receive interest gross. Authorised entities applying in the name of a Prescribed Fund will have final withholding tax (currently 10%) deducted from interest payments.
 - In terms of Section 10.2 of the Securities Note, unless the Issuer is otherwise instructed by a Bondholder, or if the Bondholder does not fall within the definition of "recipient" in terms of Article 41(c) of the Income Tax Act (Cap. 123 of the Laws of Malta), interest shall be paid to such person net of final withholding tax (currently 15%) of the gross amount of interest, pursuant to Article 33 of the Income Tax Act (Cap. 123 of the Laws of Malta).
- 8. Non-residents of Malta should note that payment of interest to individuals and certain residual entities residing in another EU Member State is reported on an annual basis to the Director General Inland Revenue, Malta, who will in turn exchange the information with the competent tax authority of the Member State where the recipient of interest is resident. This exchange of information takes place in terms of the Council Directive 2014/107/EU of 9 December 2014 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation.
- 8a. The contents of Notes 7 and 8 above do not constitute tax advice by the Issuer and Applicants are to consult their own independent tax advisors in case of doubt
- 9. Should any Application not be accepted, or be accepted for fewer Bonds than those applied for, the monies of the balance of the amount paid but not allocated, as the case may be, will be returned by direct credit into the bank account as indicated in Panel E. Interest or redemption proceeds will be credited to the account designated in Panel E or as otherwise amended by the Bondholder/s during the term of the Bond.
- 10. Completed Application Forms are to be delivered to any of the Authorised Financial Intermediaries listed in Annex II of the Securities Note during normal office hours by not later than 10:00 on 18 July 2016. Remittances by post are made at the risk of the Applicant and the Issuer disclaims all responsibility for any such remittances not being received by the closing date indicated above. The Issuer reserves the right to refuse any Application which appears to be in breach of the Terms and Conditions of the Application as contained in the Prospectus. Any Applications received by the Registrar after 10:00 on 18 July 2016 will not be accepted.
- 11. By completing and delivering an Application Form you (as the Applicant/s) acknowledge that:
 - a. the Issuer may process the personal data that you may provide in the Application Form in accordance with the Data Protection Act (Cap. 440 of the Laws of Malta);
 - b. the Issuer may process such personal data for all purposes necessary for and related to the Bonds applied for; and
 - c. you, as the Applicant, have the right to request access to and rectification of the personal data relating to you, as processed by the Issuer. Any such requests must be made in writing and addressed to the Issuer. The request must be signed by yourself, as the Applicant to whom the personal data relates.

The value of investments can go up or down and past performance is not necessarily indicative of future performance. The nominal value of the Bonds on offer will be repayable in full upon redemption. An investor should consult an independent financial adviser, licensed under the Investments Services Act (Cap. 370 of the Laws of Malta), for advice.



MIDI P.L.C. €50,000,000 4% SECURED BONDS 2026 APPLICATION FORM 'B' **GBP BONDS 2016/18**

_	Please read the notes overleaf before completing this Application Form. Ma	ark 'X' if ap	plicable.		
	APPLICANT (See notes 2 & 4)				
			TEL. NO.		MOBILE NO.
					H 0/10 / 1
	This Application Form is not transferable and entitles you to a preferential trand is to be submitted as a method of payment where the Applicant se so as to transfer to the Issuer all or part of the holding in GBP Bonds 20 which is set out in Box 1 of Panel B hereunder. For such purposes, the pa equivalent value in euro (set out in Box 2) at the exchange rate established this signed Application Form, Existing Holders shall be deemed:	lects to app 16/18 held r value of ea	ly for the Midi p.l.c. 49 by the Applicant as at ch holding in GBP Bon	6 Secure the Cut-0 ds 2016/	d Bonds 2026 (the "Bonds") Off Date, the nominal value of 18 has been converted to the
	a. to confirm that all or part (as the case may be) of the said Bonds 2 Issuer's favour for cancellation; and	016/18 as i	ndicated in the Applica	tion Forr	n are to be transferred in the
	 as an irrevocable mandate to the Issuer to engage the services of sucarry out all procedures necessary and to fully and effectively vest titled. 	ich brokers e in the app	or intermediaries as ma ropriate number of Bor	ay be ned ids in the	cessary to fully and effectively Existing Holder.
B	I/WE APPLY TO PURCHASE AND/OR ACQUIRE (See note)		•		
_	BOX 1 - Nominal Value of the GBP Bonds 2016/18.	NT IN FIGUR	RES Box 1	AMOLIN'	T IN FIGURES Box 2
	BOX 2 - Nominal Value of GBP Bonds 2016/18 converted to the equivalent value in euro at the pre-determined exchange rate of €1: £0.834, rounded up to the nearest €100.	11111001		€	THE BOXE
	AMOUNT IN WORDS			AMOUN	IT IN FIGURES Box 3
				€	
	BOX 3 - I/We wish to purchase and acquire the amount set out in Box			AMOUN	IT IN FIGURES Box 4
	Price (at par) pursuant to the Prospectus dated 28 June 2016 (minimule 100 thereafter).	ım €2,000 a	and in multiples of	Differen	ce payable on Application Box 2
	BOX 4 - Amount of Bonds applied for less the nominal holding in Bonds application under the Terms and Conditions of the Bonds set out in the F		ayable in full upon	€	
(•	(to be completed	I ONI ∨ if ti	ne Applicant is a Resident of Malta)
•	I/We elect to have final withholding tax deducted from my/our intere		(to be completed	01421 11 0	10 7 (ppilocalité lo et l'écolatorité et Marta)
	I/We elect to receive interest GROSS (i.e. without deduction of with)				
0	NON-RESIDENT - DECLARATION FOR TAX PURPOSES (See notes 3	, 7, 8 & 8a) (to be comp	oleted ONL	Y if the Applicant is a Non-Resident)
	TAX COUNTRY	CITY OF E			
	T.I.N. (Tax Identification Number) PASSPORT/NATIONAL I.D. CARD NUMBER COUN	TRY OF ISS	Y OF BIRTH UE	ISSU	E DATE
	I/We am/are NOT Resident in Malta but I/we am/are Resident in the	European l	Jnion.		
		in the Furo	oean Union		
			3341 31113111		
٦	INTEREST, REFUND & REDEMPTION MANDATE (See note	9)		Comple	etion of this Panel is MANDATORY
	BANK				
	I/We have fully understood the instructions for completing this Applic the Prospectus, and subject to its Terms and Conditions of the Bonds				
	Signature/s of Applicant/s		Dat	e	
	(Parent/s or legal guardian/s are/is to sign if Applicant is a minor) (All parties are to sign in the case of a joint Application)				
	(Bare owner/s and usufructuary/ies to sign in the case of holdings of GBP Bonds 20	016/18 that a	re subject to usufruct)		
	AUTHORISED FINANCIAL INTERMEDIARY'S STAMP AUTHORISED FII	VANCIAL IN	TERMEDIARY'S CODE		APPLICATION NUMBER

Notes on how to complete this Application Form and other information

The following notes are to be read in conjunction with the Prospectus dated 28 June 2016 regulating the Bond Issue

- 1. This Application is governed by the Terms and Conditions of the Application contained in Section 12 of the Securities Note dated 28 June 2016 forming part of the Prospectus. Capitalised terms not defined herein shall, unless the context otherwise requires, have the meaning ascribed to them in the Prospectus.
- 2. This Application Form is to be completed in BLOCK LETTERS.
- 3. Applicants who are Non-Residents in Malta for tax purposes must complete Panel D overleaf.
- 4. The MSE account number pertaining to the Applicant has been pre-printed in Panel A and reflects the MSE account number on the Issuer's Register at the CSD as at 24 June 2016 (trading session of the 22 June 2016). APPLICANTS ARE TO NOTE THAT ANY BONDS ALLOTTED TO THEM WILL BE RECORDED BY THE MALTA STOCK EXCHANGE IN THE MSE ACCOUNT QUOTED ON THIS APPLICATION FORM EVEN IF THE DETAILS OF SUCH MSE ACCOUNT, AS HELD BY THE CSD OF THE MALTA STOCK EXCHANGE, DIFFER FROM ANY OR ALL OF THE DETAILS APPEARING OVERLEAF.
- 5. In the case where a holder of Bonds 2016/18 is a body corporate, Application Forms must be signed by duly authorised representative/s indicating the capacity in which they are signing.
- 6. The amount set out in Box 3 of Panel B overleaf must be in multiples of €100. The Issuer will be giving preference to Applications made by holders of GBP Bonds 2016/18 up to their full amount held as at the Cut-Off Date converted to the equivalent value in euro at the exchange rate of €1: £0.834, rounded up to the nearest €100, as established by the Issuer on 27 June 2016. Subscriptions shall be subject to a minimum application of €2,000.

Where the Applicant wishes to acquire a number of Bonds having an aggregate value which exceeds the nominal value of the number of Bonds 2016/18 converted from GBP set out in Box 2 of Panel B, the Applicant may do so by including such higher amount in Box 3 in Panel B. In such case, the Applicant must ensure that the relative Application Form is accompanied by payment of the difference between the full price of the amount of Bonds applied for and the nominal value of Bonds 2016/18 being transferred. Payment of the amount representing such difference, which is to be inserted in Box 4 of Panel B overleaf, must be made in Euro in cleared funds to "The Registrar - Midi Bond Issue 2016". In the event that the cheque accompanying an Application Form is not honoured on the first presentation, the Issuer and the Registrar reserve the right to invalidate the relative Application.

7. Only Applicants who hold a valid official Maltese Identity Card or companies registered in Malta will be treated as residents in Malta. In such a case, the Applicant may elect to have final withholding tax, currently 15%, deducted from interest payments in which case, such interest need not be declared in the Applicant's income tax return. The Applicant may elect to receive the interest gross (i.e. without deduction of final withholding tax), but will then be obliged to declare interest so received in the tax return. Interest received by non-resident Applicants is not taxable in Malta and non-residents will receive interest gross. Authorised entities applying in the name of a Prescribed Fund will have final withholding tax (currently 10%) deducted from interest payments.

In terms of Section 10.2 of the Securities Note, unless the Issuer is otherwise instructed by a Bondholder, or if the Bondholder does not fall within the definition of "recipient" in terms of Article 41(c) of the Income Tax Act (Cap. 123 of the Laws of Malta), interest shall be paid to such person net of final withholding tax (currently 15%) of the gross amount of interest, pursuant to Article 33 of the Income Tax Act (Cap. 123 of the Laws of Malta).

- 8. Non-residents of Malta should note that payment of interest to individuals and certain residual entities residing in another EU Member State is reported on an annual basis to the Director General Inland Revenue, Malta, who will in turn exchange the information with the competent tax authority of the Member State where the recipient of interest is resident. This exchange of information takes place in terms of the Council Directive 2014/107/EU of 9 December 2014 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation.
- 8a. The contents of Notes 7 and 8 above do not constitute tax advice by the Issuer and Applicants are to consult their own independent tax advisors in case of doubt.
- 9. Should any Application not be accepted, or be accepted for fewer Bonds than those applied for, the monies of the balance of the amount paid but not allocated, as the case may be, will be returned by direct credit into the bank account as indicated in Panel E. Interest or redemption proceeds will be credited to the account designated in Panel E or as otherwise amended by the Bondholder/s during the term of the Bond.
- 10. Completed Application Forms are to be delivered to any of the Authorised Financial Intermediaries listed in Annex II of the Securities Note during normal office hours by not later than 10:00 on 18 July 2016. Remittances by post are made at the risk of the Applicant and the Issuer disclaims all responsibility for any such remittances not being received by the closing date indicated above. The Issuer reserves the right to refuse any Application which appears to be in breach of the Terms and Conditions of the Application as contained in the Prospectus. Any Applications received by the Registrar after 10:00 on 18 July 2016 will be will not be accepted.
- 11. By completing and delivering an Application Form you (as the Applicant/s) acknowledge that:
 - a. the Issuer may process the personal data that you may provide in the Application Form in accordance with the Data Protection Act (Cap. 440 of the Laws of Malta);
 - b. the Issuer may process such personal data for all purposes necessary for and related to the Bonds applied for; and
 - c. you, as the Applicant, have the right to request access to and rectification of the personal data relating to you, as processed by the Issuer. Any such requests must be made in writing and addressed to the Issuer. The request must be signed by yourself, as the Applicant to whom the personal data relates.

The value of investments can go up or down and past performance is not necessarily indicative of future performance. The nominal value of the Bonds on offer will be repayable in full upon redemption. An investor should consult an independent financial adviser, licensed under the Investments Services Act (Cap. 370 of the Laws of Malta), for advice.



MIDI p.l.c. €50,000,000 4% SECURED BONDS 2026 APPLICATION FORM 'C' **CURRENT SHAREHOLDERS**

Please read the notes overleaf before completing this Application Form. Mark 'X' if applicable.

APPLICANT (see notes 2 & 4)			
,			
E-MAIL ADDRESS		TEL. NO.	MOBILE NO.
	not transferable and entitles y		c. 4% Secured Bonds 2026 a
shareholder of Midi p.l.c.	as at the Cut-Off Date (the "Co	urrent Shareholders").	
I/WE APPLY TO PURCHA	ASE AND ACQUIRE (see notes 6	& 7)	
AMOUNT IN FIGURES	AMOUNT IN WORDS		
€			
Mid: p. Le. 40/ Consumed Date		in an object of C100 the connection	at the Denel Janua Dries (at non
	nds 2026 (minimum €2,000 and is dated 28 June 2016, (the " Pro		
Conditions of the Bonds as), payasie iii iaii aperi	application and the fermi an
	·		
DESIDENT - WITHHOLDI	NG TAX DECLARATION (See not	too 9 9 00) (to be accomplete	and ONII Viftha Amaliaantia a Daaidant of Mal
MESIDENT - WITHHOLDI	ING TAX DECLARATION (See Hol	les 8 & 9a) (to be complete	ed ONLY if the Applicant is a Resident of Mal
I/We elect to have final with	sholding tax deducted from my/our inte	erest.	
I/We elect to receive interes	st GROSS (i.e. without deduction of wi	ithholding tax).	
	,	<u> </u>	
	ARATION FOR TAX PURPOSES		pleted ONLY if the Applicant is a Non-Resider
TAX COUNTRY		TOWN OF BIRTH	
T.I.N. (Tax Identification Number) PASSPORT/NATIONAL I.D. CAR	DD NII IMBED	COUNTRY OF BIRTH	ISSUE DATE
			ISSUE DATE
I/We am/are NOT Resident	in Malta but I/we am/are Resident in t	the European Union.	
I/We am/are NOT Resident	in Malta and I/we am/are NOT Reside	ent in the European Union.	
INTEREST REFLINIR A RE			
<u> </u>	EDEMPTION MANDATE (See no	ote 10)	Completion of this Panel is MANDATOF
BANK	IBAN		
	instructions for completing this Ap	polication Form, and am/are makin	g this Application solely on the bas
I/We have fully understood the	e instructions for completing this Ap		
I/We have fully understood the			
I/We have fully understood the			
I/We have fully understood the			
I/We have fully understood the			
I/We have fully understood the			
I/We have fully understood the			
I/We have fully understood the		e Bonds (as contained therein) wh	
I/We have fully understood the of the Prospectus, and subjection of the Prospectus of the Pros	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.
I/We have fully understood the of the Prospectus, and subjection of the Prospectus of the Pr	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.
I/We have fully understood the of the Prospectus, and subjection of the Prospectus of the Pros	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.
I/We have fully understood the of the Prospectus, and subjection of the Prospectus of the Pros	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.
I/We have fully understood the of the Prospectus, and subjection of the Prospectus of the Pros	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.
I/We have fully understood the of the Prospectus, and subject the Prospectus (Prospectus Prospectus Prospec	et to its Terms and Conditions of the	e Bonds (as contained therein) wh	ich I/we fully accept.

Notes on how to complete this Application Form and other information

The following notes are to be read in conjunction with the Prospectus dated 28 June 2016 regulating the Bond Issue.

- 1. This Application is governed by the Terms and Conditions of the Application contained in Section 12 of the Securities Note dated 24 June 2016 forming part of the Prospectus. Capitalised terms not defined herein shall, unless the context otherwise requires, have the same meaning as that ascribed to them in the Prospectus.
- 2. This Application Form is to be completed in BLOCK LETTERS.
- 3. Applicants who are Non-Resident in Malta for tax purposes, must complete Panel D Overleaf.
- 4. The MSE account number pertaining to the Applicant has been preprinted in Panel A and reflects the MSE account number on the Issuer's Register at the CSD as at 24 June 2016 (trading session of the 22 June 2016). APPLICANTS ARE TO NOTE THAT ANY SECURITIES ALLOTTED TO THEM WILL BE RECORDED BY THE MALTA STOCK EXCHANGE IN THE MSE ACCOUNT NUMBER QUOTED ON THIS APPLICATION FORM EVEN IF THE DETAILS OF SUCH MSE ACCOUNT NUMBER, AS HELD BY THE CSD OF THE MALTA STOCK EXCHANGE, DIFFER FROM ANY OR ALL OF THE DETAILS APPEARING OVERLEAF.
- 5. In the case of a body corporate, the Application Form must be signed by duly authorised representative/s indicating the capacity in which they are signing.
- 6. Applications must be for a minimum of €2,000 and thereafter in multiples of €100.
- 7. Payment in Euro must be made in cleared funds to 'The Registrar Midi Bond Issue 2016'. In the event that the cheque accompanying an Application Form is not honoured on the first presentation, the Issuer and the Registrar reserve the right to invalidate the relative Application.
- 8. Only Applicants who hold a valid official Maltese Identity Card or companies registered in Malta will be treated as residents in Malta. In such a case, the Applicant may elect to have final withholding tax, currently 15%, deducted from interest payments in which case, such interest need not be declared in the Applicant's income tax return. The Applicant may elect to receive the interest gross (i.e. without deduction of final withholding tax), but will then be obliged to declare interest so received in the tax return. Interest received by Non-Resident Applicants is not taxable in Malta and Non-Residents will receive interest gross. Authorised entities applying in the name of a Prescribed Fund will have final withholding tax (currently 10%) deducted from interest payments.
 - In terms of Section 10.2 of the Securities Note, unless the Issuer is otherwise instructed by a Bondholder, or if the Bondholder does not fall within the definition of "recipient" in terms of Article 41(c) of the Income Tax Act, interest shall be paid to such person net of final withholding tax (currently 15%) of the gross amount of interest, pursuant to Article 33 of the Income Tax Act.
- 9. Non-residents of Malta should note that payment of interest to individuals and certain residual entities residing in another EU Member State is reported on an annual basis to the Director General Inland Revenue, Malta, who will in turn exchange the information with the competent tax authority of the Member State where the recipient of interest is resident. This exchange of information takes place in terms of the Council Directive 2014/107/EU of 9 December 2014 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation.
- 9a. The contents of 8 and 9 above do not constitute tax advice by the Issuer and Applicants are to consult their own independent tax advisors in case of doubt.
- 10. Should any Application not be accepted, or be accepted for fewer Bonds than those applied for, the monies of the balance of the amount paid but not allocated, as the case may be, will be returned by direct credit into the bank account as indicated in Panel E. Interests or redemption proceeds will be credited to the account designated in Panel E or as otherwise amended by the Bondholder/s during the term of the Bond.
- 11. Completed Application Forms are to be delivered to any of the Authorised Intermediaries listed in Annex II of the Securities Note during normal office hours by not later than 10:00 hours on 18 July 2016. Remittances by post are made at the risk of the Applicant and the Issuer disclaims all responsibility for any such remittances not being received by the closing date indicated above. The Issuer reserves the right to refuse any Application which appears to be in breach of the Terms and Conditions of the Application as contained in the Prospectus. Any Applications received by the Registrar after 10:00 hours on 18 July 2016 will not be accepted.
- 12. By completing and delivering an Application Form you (as the Applicant/s) acknowledge that:
 - a. the Issuer may process the personal data that you may provide in the Application Form in accordance with the Data Protection Act (Cap. 440 of the Laws of Malta);
 - b. the Issuer may process such personal data for all purposes necessary for and related to the Bonds applied for; and
 - c. you, as the Applicant, have the right to request access to and rectification of the personal data relating to you, as processed by the Issuer. Any such requests must be made in writing and addressed to the Issuer. The request must be signed by yourself, as the Applicant to whom the personal data relates.

The value of investments can go up or down and past performance is not necessarily indicative of future performance. The nominal value of the Bonds on offer will be repayable in full upon redemption. An investor should consult a licensed stockbroker or an independent financial adviser, licensed under the Investments Services Act (Cap. 370 of the Laws of Malta), for advice.



MIDI p.l.c. €50,000,000 4% SECURED BONDS 2026 APPLICATION FORM 'D' **GENERAL PUBLIC**

Please read the notes overleaf before completing this Application Form. Mark 'x' if applicable.

APPLICANT (see notes 2 to	7)		
Non-Resident	Minor (under 18)	Body Corporate/ Body of Persons	CIS-Prescribed Fund
TITLE (Mr/Mrs/Ms/)	FULL NAME AND SURNAME /	REGISTERED NAME	
ADDRESS			
			POSTCODE
MSE A/C NO. (if applicable)	I.D. CARD / PASSPORT / COI	MPANY REG. NO. TEL.	NO. MOBILE NO.
ADDITIONAL (JOINT) API	PLICANTS (see note 4)	(please use	additional Application Forms if space is not sufficient)
TITLE (Mr/Mrs/Ms/)	FULL NAME AND SURNAME	u	I.D. CARD/PASSPORT NO.
TITLE (Mr/Mrs/Ms/)	FULL NAME AND SURNAME		I.D. CARD/PASSPORT NO.
MINOR'S PARENTS / LEG	GAL GUARDIAN/S (see note 5	5)	(to be completed ONLY if the Applicant is a minor)
TITLE (Mr/Mrs/Ms/)	FULL NAME AND SURNAME	<i>'</i> 1	I.D. CARD/PASSPORT NO.
TITLE (Mr/Mrs/Ms/)	FULL NAME AND SURNAME		I.D. CARD/PASSPORT NO.
IAME APPLY TO PURCHA	SE AND ACQUIRE (see notes 8	2 & Q)·	
AMOUNT IN FIGURES	AMOUNT IN WORDS	ο α <i>θ</i>).	
€			
as defined in the Prospectus da	ited 28 June 2016 (the 'Prospectu		O thereafter) at the Bond Issue Price (at par), ation under the Terms and Conditions of the
Bonds as set out in the Prospec	tus.		
RESIDENT - WITHHOLDI	NG TAX DECLARATION (see	notes 10 & 11a) (to be	completed ONLY if the Applicant is a Resident of Malta)
	,	,	,,
/ // // vve elect to have linal with	holding tax deducted from my/our i	mieresi.	
I/We elect to receive interes	t GROSS (i.e. without deduction of	withholding tax).	
NON-RESIDENT - DECLA	RATION FOR TAX PURPOSES	(see note 3 10 11 & 11a) (fo.	be completed ONLY if the Applicant is a Non-Resident,
TAX COUNTRY	7 (110111 O11 17 01 1 O11 O0E0	CITY OF BIRTH	ре сотпрыва СпЕт и и е Аррисан из атчот т испасти,
T.I.N. (Tax Identification Number)		COUNTRY OF BIRTH	
PASSPORT/NATIONAL I.D. CAF	D NUMBER	COUNTRY OF ISSUE	ISSUE DATE
I/We am/are NOT resident	n Malta but I/we am/are resident in	the European Union	
I/We am/are NOT resident	n Malta and I/we am/are NOT resid	dent in the European Union	
INTEREST, REFUND AND	REDEMPTION MANDATE	see note 12)	Completion of this panel is MANDATORY
BANK	IBAN	,	
I/We have fully understood the	instructions for completing this	Application Form, and am/are	e making this Application solely on the basis
	t to its Terms and Conditions of t		
Signature/s of Applicant/s (Parent/s or legal guardian/s are/is to (All parties are to sign in the case of a			Date
AUTHORISED FINANCIAL INTERMEDI		NCIAL INTERMEDIARY'S CODE	APPLICATION NUMBER
		1 1 1 1	1

Notes on how to complete this Application Form and other information

The following notes are to be read in conjunction with the Prospectus dated 28 June 2016 regulating the Bond Issue

- This Application is governed by the Terms and Conditions of the Application contained in Section 12 of the Securities Note dated 28 June 2016 forming
 part of the Prospectus. Capitalised terms not defined herein shall, unless the context otherwise requires, have the meaning ascribed to them in the
 Prospectus.
- 2. The Application Form is to be completed in BLOCK LETTERS.
- 3. Applicants who are Non-Residents In Malta for tax purposes, must indicate their passport number in Panel B and complete Panel G. The relative box in Panel A must also be marked appropriately.
- 4. Applicants are to insert full personal details in Panel B. In the case of an Application by more than one person (including husband and wife) full details of all individuals, including I.D. card numbers, must be given in Panels B and C but the person whose name appears in Panel B shall, for all intents and purposes, be deemed to be the registered holder of the Bonds (vide note 7 below). Interest and redemption proceeds will be issued to the account indicated in Panel H or as otherwise indicated by the Bondholder/s during the term of the Bond.
- 5. Applications in the name and for the benefit of minors shall be allowed provided that they are signed by both parents or by the legal guardian/s and accompanied by a Public Registry birth certificate of the minor in whose name and for whose benefit the Application Form is submitted. The relative box in Panel A must also be marked appropriately. Any Bonds allocated pursuant to such an Application shall be registered in the name of the minor as Bondholder, with interest and redemption proceeds payable to the parents or legal guardian/s signing the Application Form until such time as the minor attains the age of eighteen (18) years, following which all interest and redemption proceeds shall be payable directly to the registered holder, provided that the Issuer has been duly notified in writing of the fact that the minor has attained the age of eighteen (18) years.
- 6. In the case of a body corporate, the name of the entity exactly as registered, and the registration number are to be inserted in Panel B. Applications must be signed by duly authorised representatives indicating the capacity in which they are signing.
- 7. APPLICANTS WHO ALREADY HOLD SECURITIES ON THE MSE ARE TO INDICATE THEIR MSE ACCOUNT NUMBER IN PANEL B. APPLICANTS ARE TO NOTE THAT ANY SECURITIES ALLOTTED TO THEM WILL BE RECORDED IN THE MSE ACCOUNT NUMBER QUOTED ON THIS APPLICATION FORM. IF DETAILS OF SUCH MSE ACCOUNT NUMBER, AS HELD BY THE MSE, DIFFER FROM ANY OR ALL OF THE DETAILS APPEARING OVERLEAF, A SEPARATE REQUEST BY THE APPLICANT TO CHANGE THESE DETAILS AS RECORDED AT THE MSE WILL HAVE TO BE FEFECTED.
- 8. Applications must be for a minimum of €2,000 and thereafter in multiples of €100.
- 9. Payment must be made in Euro, in cleared funds to 'The Registrar Midi Bond Issue 2016'. In the event that the cheque accompanying an Application Form is not honoured on the first presentation, the Issuer and the Registrar reserve the right to invalidate the relative Application.
- 10. Only Applicants who hold a valid official Maltese Identity Card or companies registered in Malta will be treated as resident in Malta. In such a case the Applicant may elect to have final withholding tax, currently 15%, deducted from interest payments in which case such interest need not be declared in the Applicant's income tax return. The Applicant may elect to receive the interest gross (i.e. without deduction of final withholding tax), but will be obliged to declare interest so received in the tax return. Interest received by non-resident Applicants is not taxable in Malta and non-residents will receive interest gross. Authorised entities applying in the name of a Prescribed Fund (having indicated their status in the appropriate box in Panel A) will have final withholding tax (currently 10%), deducted from interest payments.
 - In terms of Section 10.2 of the Securities Note, unless the Issuer is otherwise instructed by a Bondholder, or if the Bondholder does not fall within the definition of "recipient" in terms of Article 41(c) of the Income Tax Act (Cap. 123 of the Laws of Malta), interest shall be paid to such person net of final withholding tax,(currently 15%) of the gross amount of interest, pursuant to Article 33 of the Income Tax Act (Cap. 123 of the Laws of Malta).
- 11. Non-residents of Malta should note that payment of interest to individuals and certain residual entities residing in another EU Member State is reported on an annual basis to the Director General Inland Revenue, Malta, who will in turn exchange the information with the competent tax authority of the Member State where the recipient of interest is resident. This exchange of information takes place in terms of the Council Directive 2014/107/EU of 9 December 2014 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation.
- 11a. The contents of Notes 10 and 11 above do not constitute tax advice by the Issuer and Applicants are to consult their own independent tax advisors in case of doubt.
- 12. If any Application is not accepted after the closure of the subscription lists or is accepted for fewer Bonds than those applied for, the monies or the balance of the amount paid but not allocated, as the case may be, will be returned by direct credit into the bank account as indicated in Panel H. Interest and redemption proceeds will be credited to the account indicated in Panel H or as otherwise amended by the Bondholder/s during the term of the Bond.
- 13. Subscription lists will open at 08:30 hours on 18 July 2016 and will close as soon thereafter as may be determined by the Issuer, but not later than 16.00 hours on 20 July 2016. The Issuer reserves the right to refuse any Application which appears to be in breach of the Terms and Conditions of the Application as contained in the Prospectus. Any Applications received by the Registrar after the subscription lists close will not be accepted. Completed Application Forms are to be delivered to any of the Authorised Financial Intermediaries listed in Annex II of the Securities Note, during normal office hours. Remittances by post are made at the risk of the Applicant and the Issuer disclaims all responsibility for any such remittances not being received by the date of closing of the subscription lists.
- 14. By completing and delivering an Application Form you (as the Applicant(s)) acknowledge that:
 - a. the Issuer may process the personal data that you provide in the Application Form in accordance with the Data Protection Act (Cap. 440 of the Laws of Malta);
 - b. the Issuer may process such personal data for all purposes necessary for and related to the Bonds applied for; and
 - c. you, as the Applicant, have the right to request access to and rectification of the personal data relating to you, as processed by the Issuer. Any such requests must be made in writing and addressed to the Issuer. The request must be signed by yourself as the Applicant to whom the personal data relates

The value of investments can go up or down and past performance is not necessarily indicative of future performance. The nominal value of the Bonds on offer will be repayable in full upon redemption. An investor should consult an independent financial adviser, licensed under the Investment Services Act (Cap. 370 of the Laws of Malta), for advice.

Annex II Authorised Financial Intermediaries

Name	Address	Telephone
APS Bank Ltd	APS Centre, Tower Road, Birkirkara BKR 4012	25603000
Bank of Valletta p.l.c.	BOV Centre, Cannon Road, St Venera SVR 9030	22751732
Calamatta Cuschieri & Co Ltd	Fifth Floor, Valletta Buildings, South Street, Valletta VLT 1103	25688688
Charts Investment Management Service Ltd	Valletta Waterfront, Vault 17, Pinto Wharf, Floriana FRN 1913	21224106
Curmi & Partners Ltd	Finance House, Princess Elizabeth Street, Ta' Xbiex XBX 1102	21347331
Financial Planning Services Ltd	4, Marina Court No. 1, G. Calì Street, Ta' Xbiex XBX 1421	21344244
FINCO Treasury Management Ltd	Level 5, The Mall Complex, The Mall, Floriana FRN 1470	21220002
GlobalCapital Financial Management Ltd	Testaferrata Street, Ta'Xbiex XBX 1403	21342342
Growth Investments Ltd	Customer Service Centre, Pjazza Papa Giovanni XXIII, Floriana FRN 1420	25909357
Hogg Capital Investments Ltd	Ferris Building, Level 4, 1, St Luke's Road, Gwardamangia, Pieta PTA 1020	21322872
Jesmond Mizzi Financial Advisors Ltd	67/3, South Street, Valletta VLT 1105	23265696
Lombard Bank Malta p.l.c.	67. Republic Street, Valletta VLT 1117	25581806
Mediterranean Bank plc	10, St Barbara Bastion, Valletta VLT 1961	25574860
MFSP Financial Management Ltd	220, Immaculate Conception Street, Msida MSD 1838	21332200
Michael Grech Financial Investment Services Ltd	The Brokerage, Level O A, St Martha Street, Victoria, Gozo VCT 2550	21554492
MZ Investment Services Ltd	55, MZ House, St Rita Street, Rabat RBT 1523	21453739
Rizzo, Farrugia & Co (Stockbrokers) Ltd	Airways House, Third Floor, High Street, Sliema SLM 1549	22583000

Annex III Summary of Privileges and Hypothecs

MIDI p.l.c. - C15836 LIABILITIES

1. Cut-Off Dates

Notes of Hypothecs/Privileges: 01/01/1994 to 06/05/2016

References: 29/03/2016

2. Defined Terms

In this Report, the following capitalised have the meanings set out hereunder:

"**Bond Issue**" means the four per cent (4%) secured bonds due on 27 July 2026 being issued pursuant to the Prospectus having a nominal value of EUR100 each for an aggregate principal amount of fifty million euro (EUR50,000,000).

"BOV" means Bank of Valletta p.l.c.

"Designated Properties" means the TUD of the properties situated at Tigné Point on which MIDI shall be granting a special hypothec in favour of CSB Trustees & Fiduciaries Limited (C40390) in its capacity of Security Trustee (the Security Trustee) to secure the Bonds issued in terms of the Bond Issue.

"**Emphyteutical Deed**" means the deed in the records of Notary Vincent Miceli of the fifteenth day of June of the year two thousand (15/06/2000) entered into by the GOM and the Malta Maritime Authority of the one part and MIDI.

"Emphyteutical Site" means the TUD of Tigné Point and Manoel Island collectively.

"**GH**" means general hypothec.

"GOM" means the Government of Malta.

"HSBC" means HSBC Bank Malta p.l.c.

"Land Registry" means the registry set up by the Land Registration Act (Cap.296 of the Laws of Malta).

"Lombard" means Lombard Bank Malta p.l.c.

"Manoel Island" means the divided portion of land at Manoel Island, limits of Gżira having an area of approximately two hundred and sixty seven thousand nine hundred square metres (267,900 sq.m.) bounded on the South, East and North-East by the foreshore as shown bordered in red on the plan Land Drawing letter 'L' letter 'D' one hundred and seventy four letter 'A' bar ninety nine (LD174A/99) attached to the Emphyteutical Deed marked Schedule Two (2) including any areas occupied by streets, roads or public spaces and the buildings, structures and constructions all without number existing in, on or under the said land; and the airspace overlying the basement of the building known as The Admiralty Canteen, without number, at Manoel Island, limits of Gżira, washed in yellow on the said plan Land Drawing LD174A/99 marked Schedule Two (2) and having an area of approximately one hundred and seven decimal point five square metres (107.5sq.m.), and bounded on the North, South, and West by property granted on emphyteusis by virtue of the Emphyteutical Deed and the airspace adjacent thereto washed in blue on the said plan Land Drawing LD174A/99 marked Schedule Two (2), having an area of approximately seventy seven square metres (77sq.m.), from the level of the said basement's roof upwards and is bounded on the South by the airspace mentioned above, on the West by property granted on emphyteusis by virtue of the Emphyteutical Deed and on the East by property of the Grantor known as the Manoel Island Yacht Yard.

"MIDI" means a public company registered in Malta with registered office at North Shore, Manoel Island, limits of Gzira, Malta and company registration number letter 'C' one five eight three six (C15836).

"**Notes**" means Notes of Hypothec and Notes of Hypothec and Privilege and Notes of Reference registered in the Public Registry.

"Public Registry" means means the registry set up by the Public Registry Act (Cap.56 of the Laws of Malta).

"SH" means special hypothec.

"SP" means special privilege.

"Tigné Point" means the divided portion of land at Tigné Point, Sliema having an area of approximately one hundred and eight thousand four hundred and twenty square metres (108,420sq.m.) bounded on the South, North and South-East by the foreshore as shown bordered in red on the plan Land Drawing letter 'L' letter 'D' one hundred and seventy five letter 'A' bar ninety nine (LD175A/99) attached to the Emphyteutical Deed marked Schedule One (1) including any areas occupied by streets, roads or public spaces and the buildings, structures and constructions all without number existing in, on or under the said land.

"**TUD**" means the temporary utile dominium for a period of ninety-nine (99) years commencing from the fifteenth day of June of the year two thousand (15/06/2000), under the terms and subject to the conditions set out in the Emphyteutical Deed.

3. Interpretation

- (a) all Notes included in this Report are registered in the Public Registry of Malta.
- (b) Phase Areas referred to in the Notes included in this Report have the same meaning given to them in the Emphyteutical Deed.

4. Declaration

- 4.1 This Report does not include:
- (a) the special hypothecs on specific apartments and garages / parking spaces/ store rooms at Tigné Point granted by MIDI to purchasers during the term of promise of sale and purchase agreements signed with them as security for the repayment of advance payments made by the purchasers should they be entitled to the refund of such payments.
- (b) the general hypothecs granted by MIDI to warrant the peaceful possession of properties at Tigné Point sold or otherwise transferred by MIDI.
- 4.2 The special hypothecs mentioned in paragraph (a) are customarily cancelled by the relative purchasers upon the definitive publication of the sale and purchase of the relative properties granted as security and in any case do not include properties which shall be granted as security for the purposes of the Bond Issue.
- 4.3 The general hypothecs mentioned in (b) above secure a contingent liability should a purchaser be evicted from a property purchased or otherwise acquired. It is the custom for these hypothecs not to be taken into account when banking facilities are provided by banks and other financial institutions.

LIST OF EFFECTIVE HYPOTHECS AND PRIVILEGES

1	
Note Number	9973
Year	2000
Creditor	GOM
Credit 1	Annual and temporary Ground-rent of: EUR1,118,100 formerly Lm480,000 per annum for the period from the 15th June 2000 to the 31/03/2025; EUR1,956,673 formerly Lm840,000 per annum for the period commencing on the first 1st April 2025 and ending on the 31st March 2050; and EUR2,236,198 formerly Lm960,000 per annum for the period commencing on the 1st April 2050 and ending on the date of expiration of the duration of the Emphyteutical Grant.
Credit 2	Payment of Premium amounting to EUR91,707,431 formerly Lm39,370,000, payable without interest in the manner stated in the Note and the Emphyteutical Deed. Reduced to EUR52,176,215.17 pursuant to several reductions the last one being in the Records of Notary Pierre Attard of the 10th May 2016.
Cause of Preference	
GH	Yes, but only in respect of Credit 1.
SH	None
SP	On the Emphyteutical Site
Notary	Vincent Miceli
Date of Deed	15/06/2000
References	
R.472/2007**	Credit 2 reduced to EUR78,732,820.87 formerly Lm33,800,000 and Release, in respect of Credit 2 only, of the TUD of the site at Tigné Point occupied by Complex A also known as Caravaggio Court comprising the 4 blocks of apartments known as T4F, T4B, T5F, T5B and the underlying basements consisting of lock up garages, garage spaces and store rooms and the relative common parts of the blocks, the basements and the entire complex and the airspace thereof.
R.9144/2007**	Credit 2 reduced to EUR75,331,936 and Release, in respect of Credit 2 only, of the TUD of the site at Tigné Point occupied by Complex B also known as Favray Court comprising the 4 blocks of apartments known as T6F, T6B, T7F, T7B and the underlying basements consisting of lock up garages, garage spaces and store rooms and the relative common parts of the blocks, the basements and the entire complex and the airspace thereof.
R. 792/2009**	Credit 2 reduced to EUR72,187,281.62 and Release, in respect of Credit 2 only, of the TUD of: (a) the site at Tigné Point occupied by the blocks of apartments known as T8F, T8B, T9F, T9B and the underlying basements consisting of lock up garages, garage spaces and store rooms; (b) the lock-up garages known as D120, D212, D225, D230, D312, D325, D330 situate at Level 4.5 of the basement with their relative common parts of the basement which underlies the Complex D also known as T10 at Tigné Point, which garages form part of the garages known as the T10 Garages.
R.793/2009	Credit 2 reduced to EUR74,167,249 and Release, in respect of Credit 2 only, of the TUD of: (a) Tigné South Phase Area - 21,141sq.m. at Tigné Point; and (b) Tigné Sports Phase Area - 14,102sq.m.at Tigné Point.

R.5221/2010**

Credit 2 reduced by EUR2,204,306.20 to EUR69,982,975.43 and Release, in respect of Credit 2 only, of the TUD of:

(a) the building known as T10F with the relative common parts and rights and appurtenances situated in the Tigné North Three and Fort Phase Area at Tigné Point; and

(b) the lock-up garages known as D116, D117, D121, D122, D123, D124, D125, D126, D127, D129, D130, D131, D132, D133, D134, D136, D201, D202, D215, D220, D221, D222, D223, D224, D226, D227, D228, D229, D239, D240, D320, D321, D328, D329 situate at Level 3 of the basement with their relative common parts of the basement which underlies the Complex D also known as T10 at Tigné Point, which garages form part of the garages known as the T10 Garages.

R.8160/2010

Credit 2 reduced to EUR66,019,939.44 and Release, in respect of Credit 2 only, of the TUD of the site (including structures thereon) - 47,376sq.m. known as the Tigné North Three and Fort Phase Area at Tigné Point.

R.8161/2010**

Release, in respect of Credit 2 only, of the TUD of the site – 20sq.m forming part of the TUD of the Tigné North Two Phase Area at Tigné Point on which Block T10B is built.

R.2830/2012**

Credit 2 reduced to EUR65,051,218.69 and Release, in respect of Credit 2 only, of the TUD of: (a) seven apartments numbered 1 through 7 within T4P at Pjazza Tigné at Tigné Point with their relative common parts;

(b) six apartments numbered 8 through 13 within T7P at Pjazza Tigné at Tigné Point with their relative common parts;

(c) nine apartments numbered 14 through 22 within T9P at Pjazza Tigné at Tigné Point with their relative common parts;

(d) forty nine garage spaces on level -3 (+4.465 above sea level) numbered P301 through P349 within car park at Tigné Point with their relative common parts; and

(e) nine store rooms (listed in Document B attached to the deed) at various levels within the buildings T4P, T7P and T9P at Tigné Point with their relative common parts.

R.2831/2012**

Credit 2 reduced to EUR65,039,750.44 and Release, in respect of Credit 2 only, of the TUD of the Tigné North Two Phase Area at Tigné Point on which part of the building T9P is built.

R.1921/2014**

Credit 2 reduced to EUR59,002,595.20 and Release, in respect of Credit 2 only, of the TUD of the divided portion of land at Tigné Point – 7945sq.m. known as the Tigné North Two Phase Area.

R.4703/2014

Clarification re Credit 1 that when the parcels of land forming part of the Tigné North Two Phase Area intended for the construction of the building known as T14 is transferred the transferee shall be liable only of the ground-rent apportioned on the parcel of land in accordance with the Utilisable Floorspace.

R.2809/2015

Apportionment of ground-rent in respect of apartments within the building known as Q1 (formerly T17E) at Tigné Point and in respect of garages, motor cycle bays and store rooms in the underground car park on the 1st, 2nd, 3rd and 4th basement levels underlying the Q1 building.

R. 3381/2016 Not. Pierre Attard 10/05/16 Credit 2 reduced to EUR52,176,215.17 and Release, in respect of Credit 2 only, of the TUD of the divided portion of land at Tigné Point, known as the Tigné Tower Phase Area - 5,760sq.m. including the structures thereon.

R. 3382 /2016 Not. Pierre Attard 10/05/16 Release, in respect of Credit 2 only, of the TUD of the divided portion of land at Tigné Point, known as the Tigné Plaza Phase Area -7,769sq.m. including the structures thereon.

R. 3383/2016** Not. Pierre Attard 10/05/16

Release, in respect of Credit 2 only, of the TUD of:

(i) the apartments (to the extent that they are situated within the Tigné North One Phase Area), namely sixty (60) apartments internally numbered one (1) through sixty (60) situated with the building known as Q2 formerly known T17W or T17 situated at Tigné Point, with their relative common parts and rights and appurtenances;

(ii) the ground floor premises (to the extent that they are situated within the Tigné North One Phase Area), namely unit A and unit B forming part of the building known as "Q2" situated at Tigné Point, with their relative common parts and rights and appurtenances;

(iii) The garages and parking spaces (to the extent that they are situated within the Tigné North One Phase Area) unofficially numbered Q126, Q130 through Q141, Q144, Q145, Q242, Q244, Q245, Q247, Q356, Q358, Q401 through Q407 with their relative common parts and rights and appurtenances, situated within the car park, known as the "Q Car Park" which Car Park underlies the Q2, the business centre, the north shore garden, and a block known Q1, at Tigné Point;

(iv) the motor cycle bays (to the extent that they are situated within the Tigné North One Phase Area) unofficially numbered letter M.C.1A and M.C.1B, with their relative common parts and rights and appurtenances, situated within the Q Car Park; and

(v) the store rooms (to the extent that they are situated within the Tigné North One Phase Area) unofficially numbered SQ104 through SQ106 and SQ207 through SQ210, with their relative common parts and rights and appurtenances, situated within the Q Car Park.

R. 3384 /2016 Not. Pierre Attard 10/05/16

Release, in respect of Credit 2 only, of the TUD of the parking spaces unofficially numbered N401 through N420, situated within the Tigné North One Phase Area, together with all their rights and appurtenances including their proportionate share pro *indiviso* of the common parts and not including their airspace or sub-terrain, situated within the car park at Tigné Point in Sliema known as the T15 Car Park situated on the fourth (4th) basement level at Tigné Point.

**

In the Notes of Reference indicated with ** it is stated that when a Unit is transferred to a Transferee (as defined in Clause Eight (8) of Schedule Twenty One (21) of the Emphyteutical Deed) the special privilege mentioned in the said Note of Hypothec and Privilege and Land Registry charge, to the extent only that the special privilege was registered to secure the payment of the ground-rent, shall continue to attach to such Unit only for the Apportioned Ground-rent as agreed by the Grantor and the Emphyteuta earlier on this deed, it being understood that the effects of this paragraph for each Unit shall be suspended until such time when the Transfer of a Unit is made to a Transferee and shall come into effect immediately and without the need of any further formality in relation to a Unit when such Unit is Transferred;

Pending Action

No pending action

-	
_	
-	

 Note Number
 20642

 Year
 2006

 Creditor
 BOV

Credit

Extension of Overdraft Lm800,000 (EUR 1,863,498.70)

Cause of Preference GH

Yes - Excluding TUD of the sites at Tigné Point known as:

T10 (phase 2)
T4 (phase 1)
T5 (phase 1)
T6 (phase 1)
T7 (phase 1)
T8 (phase 1)
T9 (phase 1)

SH TUD of:

1. The site – 5,000sq.m. at Tigné Point known as Tigné Plaza (Pjazza Tigné – T11) and all improvements to be constructed thereon consisting of the pedestrian surface and a 4 storey underground car park.

2. The site - 2,500sq.m. at Tigné Point and all improvements thereon 3 blocks to be known as T4P, T7P and T9P

SP None

Notary Marco Farrugia
Date of Deed 08/11/2006

References

R. 2048/2014 Postponement in favour of GOM H.4768/2014 up to the amount of EUR6,056,371 in so far as

GH affects divided portions of the T14 Site including North Shore Garden.

R. 4661/2014 Postponement in favour of GOM H.4768/2014 and H.11,446/2014 up to the amount of

EUR6,056,371 in so far as GH affects divided portions of the T14 Site.

Pending Action To be Cancelled - against provision of cash collateral to the Creditor of EUR2,200,000 as set

out in the Securities Note.

3	
Note Number	19686
Year	2009
Creditor	Lombard

Credit Loan EUR 4,000,000

Cause of Preference

GH Yes

SH The TUD of divided portions of land at Manoel Island Gzira known as The Marina South

Phase Area, the Marina Central Phase Area, the Manoel Sports Club Phase Area;

SP None

Notary Marco Farrugia
Date of Deed 23/12/2009

References Postponements in favour of H.1950/2010 in respect only of the TUD of:

R.1087/2010 (a) parcel of land – 5,000sq.m. at Tigné Point known as Tigné Plaza or Pjazza Tigné

consisting of a pedestrian surface and a four storey underground car park;

(b) parcel of land – 2,500sq.m. at Tigné Point and improvements thereon intended for the

construction of blocks known as T4P, T7P and T9P; and

(c) parcel of land – 2,217sq.m. at Tigné Point intended for the construction of the block

known as T20.

R.3143/2010 Postponements in favour of H.6493/2010 in respect only of the TUD of the residential block

at Tigné Point known as T10 - footprint -2700sq.m.

R.5497/2010 Release of the TUD of apartment 6 and parking spaces D117 and D201 forming part of T10F

at Tigné Point.

R.7764/2010 Postponements in favour of H.6765/2010 in respect only of the TUD of:

(a) parcel of land – 5,000sq.m. at Tigné Point known as Tigné Plaza or Pjazza Tigné or T11

consisting of a pedestrian surface and a four storey underground car park;

(b) parcel of land – 2,500sq.m. at Tigné Point and improvements thereon intended for the

construction of blocks known as T4P, T7P and T9P; and

(c) parcel of land – 2,217sq.m. at Tigné Point intended for the construction of the block

known as T20.

Release of the TUD of apartments numbered 3, 21, 23, 25, 27, 31, 35, 39, 43, 45, 47, 51, 54, 56 R.8836/2010 and 58 forming part of the complex known as T10 at Tigné Point. R.1223/2011 Release of the TUD of the apartment 40 and the lock-up garage D104 forming part of block known as T10B at Tigné Point. Release of the TUD of R.8282/2011 (a) the apartments 21, 25, 38, 40, 49, 51, 52, 54, 58 in the block known as T10B at Tigné Point: and (b) the apartments 6, 7 and 10 in the block known as T10B at Tigné Point; Release of the TUD of the following properties at Tigné Point: R.8018/2012 the parcel of land known as T17E - 640sq.m. the parcel of land known as T17W - 910sq.m. the office block or site known as T14 - 2,025sq.m. R.5075/2013 Release of the TUD of divided portion of land – 3,308sq.m. known as T12 and also as the Clubhouse at Tigné Point. R.2050/2014 Release of the TUD of the T14 Site. R.4607/2014 Release of the TUD of the T14 Site - registered to correct description of T14 Site. R. 2917/2015 Release of TUD of divided portion of land known as Q1 formerly T17E - 640sq.m. at Tigné Point. R.7109/2015 Release of TUD two commercial outlets at Tigné Point known as: RU01 also known as T4P-01: and RU02 also known as T4P -02. **Pending Action** To be Reduced - Waiver of GH to release Designated Properties situated at Tigné Point. **Note Number** 1950 Year 2010 Creditor ROV/ Credit Extension of Overdraft EUR1,146,362 Cause of Preference GH Yes - Excluding TUD of the sites at Tigné Point known as T10 (Phase 2) including above and below ground. SH TUD of: 1 The site - 5,000sq.m. at Tigné Point known as Tigné Plaza (Pjazza Tigné - T11) and all improvements to be constructed thereon consisting of the pedestrian surface and a 4 storey underground car park. 2. The site - 2,500sq.m. at Tigné Point and all improvements thereon 3 blocks to be known as T4P. T7P and T9P. 3. The site – 2,217sq.m. known as Block T20 (For EUR 3,242,798)

SP None

Notary Marco Farrugia
Date of Deed 29/01/2010

References

R. 2048/2014 Postponement in favour of GOM H.4768/2014 up to the amount of EUR6,056,371 in so far

as GH affects divided portions of the T14 Site including North Shore Garden.

R. 4661 / 2014 Postponement in favour of GOM H.4768/2014 and H.11,446/2014 up to the amount of

EUR6,056,371 in so far as GH affects divided portions of the T14 Site.

Pending Action To be Cancelled - against provision of cash collateral to the Creditor of EUR2,200,000 as set

out in the Securities Note.

Note Number 8295
Year 2012

Credit Loan EUR1,500,000

Lombard

Cause of Preference

Creditor

GH Yes

SH 1. TUD of the divided portion of land at Manoel Island Gzira known as The Marina South

Phase Area, the Marina Central Phase Area, the Manoel Sports Club Phase Area; and

2. TUD T12 also known as the Club House, Tigné, Sliema.

SP None

Notary Sam Abela
Date of Deed 23/05/2012

Undertakings MIDI undertook not to grant further hypothecary charges on the properties described above,

even if these rank after the charges in favour of the Creditor and / or not to lease, sublet or

allow third parties to use the said properties, under any title whatsoever.

References

R.8018/2012 Release of the TUD of the following properties at Tigné Point:

the parcel of land known as T17E - 640sq.m; the parcel of land known as T17W - 910sq.m; the office block or site known as T14 - 2025sq.m;

R.5075/2013 Release of the TUD of divided portion of land – 3,308sq.m. known as T12 and also as the

Clubhouse at Tigné Point.

R.2050/2014 Release of the TUD of the T14 Site.

R.4607/2014 Release of the TUD of the T14 Site – registered to correct description of T14 Site.

R. 2917/2015 Release of TUD of divided portion of land known as Q1 formerly T17E – 640sq.m. at Tigné Point.

R.7109/2015 Release of TUD two commercial outlets at Tigné Point known as:

RU01 also known as T4P-01; and RU02 also known as T4P-02.

Pending Action To be Reduced - Waiver of GH to release Designated Properties situated at Tigné Point.

Note Number 8296 Year 2012

Credit Loan EUR1,000,000

Lombard

Cause of Preference

Creditor

GH Yes

SH 1. TUD divided portion of lands at Manoel Island Gzira known as The Marina South Phase

Area, the Marina Central Phase Area, the Manoel Sports Club Phase Area; and

2. TUD T12 also known as the Club House, Tigné, Sliema.

SPNoneNotarySam AbelaDate of Deed23/05/2012

Undertakings MIDI undertook not to grant further hypothecary charges on the properties described above,

even if these rank after the charges in favour of the Creditor and / or not to lease, sublet or

allow third parties to use the said properties, under any title whatsoever.

References

R.8018/2012 Release of the TUD of the following properties at Tigné Point:

the parcel of land known as T17E - 640sq.m; the parcel of land known as T17W - 910sq.m; the office block or site known as T14 - 2,025sq.m;

R.5075/2013 Release of the TUD of divided portion of land – 3,308sq.m. known as T12 and also as the

Clubhouse at Tigné Point.

R.2050/2014 Release of the TUD of the T14 Site;

R.4607/2014 Release of the TUD of the T14 Site – registered to correct description of T14 Site;

R. 2917/2015 Release of TUD of divided portion of land known as Q1 formerly T17E – 640sq.m. at Tigné Point.

R.7109/2015 Release of TUD two commercial outlets at Tigné Point known as:

RU01 also known as T4P-01, and RU02 also known as T4P-02.

Pending Action To be Reduced - Waiver of GH to release Designated Properties situated at Tigné Point.

Note Number 4768

Year 2014 Creditor GOM

Credit Additional Security for Payment of Premium EUR6,037,155.24

Cause of Preference

GH None

SH TUD on site at Tigné Point known as T14 – 2,025sq.m. from 16.5 meters above sea level.

SP None

Notary Peter Fleri Soler

Date of Deed 20/03/2014

Pending Action No Pending Action – property burdened is not part of the Designated Properties.

8

Note Number 11446 Year 2014 Creditor GOM

Credit Additional Security for Payment of Premium EUR6,037,155.24

Remarks Registered to amend description of site T14 as described in H.4768/2014.

Cause of Preference

GH None

SH TUD on part of the site at Tigné Point known as T14

SP None

Notary Pierre Attard

Date of Deed 11/07/2014

Pending Action No Pending Action – property burdened is not part of the Designated Properties.

,

Note Number 11685 Year 2014 Creditor Mid Knight Holdings Limited

Credit Warranty for Peaceful Possession and other representations, warranties and guarantees in

respect of the sale of the T14 Site for the price of EUR11,700,000.

Cause of Preference

GH Yes

SH None - Creditor renounced its right under Section 2016 of the Civil Code to register as further

security a special hypothec on the property of MIDI.

SP None

Pierre Attard **Notary Date of Deed** 12/07/2014 **Pending Action**

No Pending Action

Note Number 20137 Year 2015

Creditor Mediterranean Corporate Bank Limited

Credit Loan EUR13,000,000

Cause of Preference

GH Yes but Excluding TUD of site T17E also known as Q1- 640sq.m.

SH TUD of site at Tigné Point known as T17W and also known as Q2 - 910sq.m. SP TUD of site at Tigné Point known as T17W and also known as Q2 - 910sq.m.

Notary Tiziana Maria Refalo

Date of Deed 28/11/2014

Undertakings

MIDI undertook not to give without the Creditor's prior written consent any further hypothecs on its property including T17W, even if these rank after the hypothecs/charges registered in favour of the Creditor and not to transfer, not to let, not to part with and not to allow third parties to use and all of the T17W under any title whatsoever, without the Creditor's prior written consent.

Pending Action

1. To be Reduced - Creditor to postpone its rights arising from the GH in so far as these affect the Designated Properties in favour of the Security Trustee - see letter dated 24th May 2016.

2. To be Reduced - Creditor to waive its hypothecary charges in so far as these affect only the commercial units situated at the ground floor (and their respective common parts) within T17 West, also known as Q2 development block, in Tigné Point, whilst keeping all its other rights firm, valid and unimpaired - see letter dated 24th May 2016.

Note Number 7568 Year 2015 Creditor BOV

Credit Additional Security for Overdraft EUR2,432,940

Cause of Preference

GH None SH TUD of site – 3,308sq.m. at Tigné Point known as T12 and also known as the Clubhouse

including the two restaurants known as The Chophouse and La Cucina del Sole, an outside

poolside kiosk, a gymnasium.

SP None

Notary Sam Abela Date of Deed 27/05/2015

Pending Action To be Cancelled - against provision of cash collateral to the Creditor of EUR2,200,000 as set

out in the Securities Note.

12

Note Number 22354
Year 2015
Creditor HSBC

creditAdditional Security Overdraft EUR791,987Principal DebtorSolutions & Infrastructure Services Limited

Surety MIDI

Cause of Preference

GH YES

SH TUD of two commercial outlets at Tigné Point known as:

RU01 also known as T4P-01; and RU02 also known as T4P -02.

SP None

Notary Michael Galea

Date of Deed 02/12/2015

Pending Action To be Cancelled - against full repayment of loan and overdraft facilities as set out in the

Securities Note.

13

Note Number 22355
Year 2015
Creditor HSBC

Credit Additional Security Loan EUR728,683.48

Principal Debtor Solutions & Infrastructure Services Limited

Surety MIDI

Cause of Preference

GH YES

SH TUD of two commercial outlets at Tigné Point known as:

RU01 also known as T4P-01; and RU02 also known as T4P -02.

SP None

Notary Michael Galea

Date of Deed 02/12/2015

Pending Action To be Cancelled - against full repayment of loan and overdraft facilities as set out in the

Securities Note.

14 **Note Number** 6743 Year 2016 Creditor Lombard Credit Loan EUR6,500,000 Cause of Preference GH Yes SH TUD divided portion of lands at Manoel Island Gzira known as The Marina South Phase Area, the Marina Central Phase Area, the Manoel Sports Club Phase Area. SP None Tiziana Maria Refalo **Notary**

Date of Deed 14/04/2016
Undertakings To be Reduced - Waiver of GH to release Designated Properties situated at Tigné Point - see

Lombard letter dated 26th May 2016.